CONTRACT SERVICES AGREEMENT

By and Between

CITY OF CARSON

and

ACTIVE NETWORK, LLC

AGREEMENT FOR CONTRACT SERVICES BETWEEN THE CITY OF CARSON AND ACTIVE NETWORK, LLC

THIS AGREEMENT FOR CONTRACT SERVICES (herein "Agreement") is made and entered into this 16th day of March, 2021 by and between the City of Carson, a California municipal corporation ("City") and Active Network, LLC, a Delaware Limited Liability Company ("Consultant"). City and Consultant may be referred to, individually or collectively, as "Party" or "Parties."

RECITALS

- A. City has sought, by issuance of a Request for Proposals or Invitation for Bids, the performance of the services defined and described particularly in Article 1 of this Agreement.
- B. Consultant, following submission of a proposal or bid for the performance of the services defined and described particularly in Article 1 of this Agreement, was selected by the City to perform those services.
- C. Pursuant to the City of Carson's Municipal Code, City has authority to enter into and execute this Agreement.
- D. The Parties desire to formalize the selection of Consultant for performance of those services defined and described particularly in Article 1 of this Agreement and desire that the terms of that performance be as particularly defined and described herein.

OPERATIVE PROVISIONS

NOW, THEREFORE, in consideration of the mutual promises and covenants made by the Parties and contained herein and other consideration, the value and adequacy of which are hereby acknowledged, the parties agree as follows:

ARTICLE 1. SERVICES OF CONSULTANT

1.1 <u>Scope of Services.</u>

In compliance with all terms and conditions of this Agreement, the Consultant shall provide those services specified in the "Scope of Services" attached hereto as <u>Exhibit "A"</u> and incorporated herein by this reference, which may be referred to herein as the "services" or "work" hereunder. As a material inducement to the City entering into this Agreement, Consultant represents and warrants that it has the qualifications, experience, and facilities necessary to properly perform the services required under this Agreement in a thorough, competent, and professional manner, and is experienced in performing the work and services contemplated herein. Consultant shall at all times faithfully, competently and to the best of its ability, experience and talent, perform all services described herein. Consultant covenants that it shall follow the highest professional standards in performing the work and services required hereunder and that all materials will be both of good quality as well as fit for the purpose intended. For purposes of this Agreement, the phrase "highest

professional standards" shall mean those standards of practice recognized by one or more first-class firms performing similar work under similar circumstances.

1.2 <u>Consultant's Proposal.</u>

The Scope of Service shall include the Consultant's scope of work or bid which shall be incorporated herein by this reference as though fully set forth herein. In the event of any inconsistency between the terms of such proposal and this Agreement, the terms of this Agreement shall govern.

1.3 <u>Compliance with Law.</u>

Consultant shall keep itself informed concerning, and shall render all services hereunder in accordance with, all ordinances, resolutions, statutes, rules, and regulations of the City and any Federal, State or local governmental entity having jurisdiction in effect at the time service is rendered.

1.4 Licenses, Permits, Fees and Assessments.

Consultant shall obtain at its sole cost and expense such licenses, permits and approvals as may be required by law for the performance of the services required by this Agreement. Consultant shall have the sole obligation to pay for any fees, assessments and taxes, plus applicable penalties and interest, which may be imposed by law and arise from or are necessary for the Consultant's performance of the services required by this Agreement, and shall indemnify, defend and hold harmless City, its officers, employees or agents of City, against any such fees, assessments, taxes, penalties or interest levied, assessed or imposed against City hereunder.

1.5 <u>Familiarity with Work.</u>

By executing this Agreement, Consultant warrants that Consultant (i) has thoroughly investigated and considered the scope of services to be performed, (ii) has carefully considered how the services should be performed, and (iii) fully understands the facilities, difficulties and restrictions attending performance of the services under this Agreement. If the services involve work upon any site, Consultant warrants that Consultant has or will investigate the site and is or will be fully acquainted with the conditions there existing, prior to commencement of services hereunder. Should the Consultant discover any latent or unknown conditions, which will materially affect the performance of the services hereunder, Consultant shall immediately inform the City of such fact and shall not proceed except at Consultant's risk until written instructions are received from the Contract Officer.

1.6 Care of Work.

The Consultant shall adopt reasonable methods during the life of the Agreement to furnish continuous protection to the work, and the equipment, materials, papers, documents, plans, studies and/or other components thereof to prevent losses or damages, and shall be responsible for all such damages, to persons or property, until acceptance of the work by City, except such losses or damages as may be caused by City's own negligence.

01007,0001/544964.15 -2-

1.7 Further Responsibilities of Parties.

Both parties agree to use reasonable care and diligence to perform their respective obligations under this Agreement. Both parties agree to act in good faith to execute all instruments, prepare all documents and take all actions as may be reasonably necessary to carry out the purposes of this Agreement. Unless hereafter specified, neither party shall be responsible for the service of the other.

1.8 <u>Additional Services.</u>

City shall have the right at any time during the performance of the services, without invalidating this Agreement, to order extra work beyond that specified in the Scope of Services or make changes by altering, adding to or deducting from said work. No such extra work may be undertaken unless a written order is first given by the Contract Officer to the Consultant, incorporating therein any adjustment in (i) the Contract Sum for the actual costs of the extra work, and/or (ii) the time to perform this Agreement, which said adjustments are subject to the written approval of the Consultant. Any increase in compensation of up to ten percent (10%) of the Contract Sum or \$25,000, whichever is less; or, in the time to perform of up to one hundred eighty (180) days, may be approved by the Contract Officer. Any greater increases, taken either separately or cumulatively, must be approved by the City Council. It is expressly understood by Consultant that the provisions of this Section shall not apply to services specifically set forth in the Scope of Services. Consultant hereby acknowledges that it accepts the risk that the services to be provided pursuant to the Scope of Services may be more costly or time consuming than Consultant anticipates and that Consultant shall not be entitled to additional compensation therefor. City may in its sole and absolute discretion have similar work done by other contractors. No claims for an increase in the Contract Sum or time for performance shall be valid unless the procedures established in this Section are followed.

1.9 Special Requirements.

Additional terms and conditions of this Agreement, if any, which are made a part hereof are set forth in the "Special Requirements" attached hereto as <u>Exhibit "B"</u> and incorporated herein by this reference. In the event of a conflict between the provisions of <u>Exhibit "B"</u> and any other provisions of this Agreement, the provisions of Exhibit "B" shall govern.

ARTICLE 2. COMPENSATION AND METHOD OF PAYMENT.

2.1 <u>Contract Sum.</u>

Subject to any limitations set forth in this Agreement, City agrees to pay Consultant the amounts specified in the "Schedule of Compensation" attached hereto as <u>Exhibit "C"</u> and incorporated herein by this reference. The total compensation, including reimbursement for actual expenses, shall not exceed Two Hundred Fifty Thousand Dollars (\$250,000.00) (the "Contract Sum"), unless additional compensation is approved pursuant to Section 1.8.

01007,0001/544964.15 -3-

2.2 <u>Method of Compensation.</u>

The method of compensation may include: (i) a lump sum payment upon completion; (ii) payment in accordance with specified tasks or the percentage of completion of the services, less contract retention; (iii) payment for time and materials based upon the Consultant's rates as specified in the Schedule of Compensation, provided that (a) time estimates are provided for the performance of sub tasks, (b) contract retention is maintained, and (c) the Contract Sum is not exceeded; or (iv) such other methods as may be specified in the Schedule of Compensation.

2.3 Reimbursable Expenses.

Compensation may include reimbursement for actual and necessary expenditures for reproduction costs, telephone expenses, and travel expenses approved by the Contract Officer in advance, or actual subcontractor expenses of an approved subcontractor pursuant to Section 4.5, and only if specified in the Schedule of Compensation. The Contract Sum shall include the attendance of Consultant at all project meetings reasonably deemed necessary by the City. Coordination of the performance of the work with City is a critical component of the services. If Consultant is required to attend additional meetings to facilitate such coordination, Consultant shall not be entitled to any additional compensation for attending said meetings.

2.4 Invoices.

Each month Consultant shall furnish to City an original invoice for all work performed and expenses incurred during the preceding month in a form approved by City's Director of Finance. By submitting an invoice for payment under this Agreement, Consultant is certifying compliance with all provisions of the Agreement. The invoice shall detail charges for all necessary and actual expenses by the following categories: labor (by sub-category), travel, materials, equipment, supplies, and subcontractor contracts. Subcontractor charges shall also be detailed by such categories. Consultant shall not invoice City for any duplicate services performed by more than one person.

City shall independently review each invoice submitted by the Consultant to determine whether the work performed and expenses incurred are in compliance with the provisions of this Agreement. Except as to any charges for work performed or expenses incurred by Consultant which are disputed by City, or as provided in Section 7.3, City will use its best efforts to cause Consultant to be paid within forty-five (45) days of receipt of Consultant's correct and undisputed invoice; however, Consultant acknowledges and agrees that due to City warrant run procedures, the City cannot guarantee that payment will occur within this time period. In the event any charges or expenses are disputed by City, the original invoice shall be returned by City to Consultant for correction and resubmission. Review and payment by City for any invoice provided by the Consultant shall not constitute a waiver of any rights or remedies provided herein or any applicable law.

2.5 Waiver.

Payment to Consultant for work performed pursuant to this Agreement shall not be deemed to waive any defects in work performed by Consultant.

01007,0001/544964.15 -4-

ARTICLE 3. PERFORMANCE SCHEDULE

3.1 Time of Essence.

Time is of the essence in the performance of this Agreement.

3.2 <u>Schedule of Performance.</u>

Consultant shall commence the services pursuant to this Agreement upon receipt of a written notice to proceed and shall perform all services within the time period(s) established in the "Schedule of Performance" attached hereto as <u>Exhibit "D"</u> and incorporated herein by this reference. When requested by the Consultant, extensions to the time period(s) specified in the Schedule of Performance may be approved in writing by the Contract Officer but not exceeding one hundred eighty (180) days cumulatively.

3.3 Force Majeure.

The time period(s) specified in the Schedule of Performance for performance of the services rendered pursuant to this Agreement shall be extended because of any delays due to unforeseeable causes beyond the control and without the fault or negligence of the Consultant, including, but not restricted to, acts of God or of the public enemy, unusually severe weather, fires, earthquakes, floods, epidemics, quarantine restrictions, riots, strikes, freight embargoes, wars, litigation, and/or acts of any governmental agency, including the City, if the Consultant shall within ten (10) days of the commencement of such delay notify the Contract Officer in writing of the causes of the delay. The Contract Officer shall ascertain the facts and the extent of delay, and extend the time for performing the services for the period of the enforced delay when and if in the judgment of the Contract Officer such delay is justified. The Contract Officer's determination shall be final and conclusive upon the parties to this Agreement. In no event shall Consultant be entitled to recover damages against the City for any delay in the performance of this Agreement, however caused, Consultant's sole remedy being extension of the Agreement pursuant to this Section.

3.4 Term.

Unless earlier terminated in accordance with Article 7 of this Agreement, this Agreement shall continue in full force and effect until completion of the services but not exceeding one (1) years from the date hereof, except as otherwise provided in the Schedule of Performance (Exhibit "D").

ARTICLE 4. COORDINATION OF WORK

4.1 Representatives and Personnel of Consultant.

The following principals of Consultant ("Principals") are hereby designated as being the principals and representatives of Consultant authorized to act in its behalf with respect to the work specified herein and make all decisions in connection therewith:

(Name	e) (Title))

(Name)	(Title)
(Name)	(Title)

It is expressly understood that the experience, knowledge, capability and reputation of the foregoing principals were a substantial inducement for City to enter into this Agreement. Therefore, the foregoing principals shall be responsible during the term of this Agreement for directing all activities of Consultant and devoting sufficient time to personally supervise the services hereunder. All personnel of Consultant, and any authorized agents, shall at all times be under the exclusive direction and control of the Principals. For purposes of this Agreement, the foregoing Principals may not be replaced nor may their responsibilities be substantially reduced by Consultant without the express written approval of City. Additionally, Consultant shall utilize only competent personnel to perform services pursuant to this Agreement. Consultant shall make every reasonable effort to maintain the stability and continuity of Consultant's staff and subcontractors, if any, assigned to perform the services required under this Agreement. Consultant shall notify City of any changes in Consultant's staff and subcontractors, if any, assigned to perform the services required under this Agreement, prior to and during any such performance.

4.2 <u>Status of Consultant.</u>

Consultant shall have no authority to bind City in any manner, or to incur any obligation, debt or liability of any kind on behalf of or against City, whether by contract or otherwise, unless such authority is expressly conferred under this Agreement or is otherwise expressly conferred in writing by City. Consultant shall not at any time or in any manner represent that Consultant or any of Consultant's officers, employees, or agents are in any manner officials, officers, employees or agents of City. Neither Consultant, nor any of Consultant's officers, employees or agents, shall obtain any rights to retirement, health care or any other benefits which may otherwise accrue to City's employees. Consultant expressly waives any claim Consultant may have to any such rights.

4.3 Contract Officer.

The Contract Officer shall be Bobby Grove, Recreation Program Manager, or such other person as may be designated by the City Manager. It shall be the Consultant's responsibility to assure that the Contract Officer is kept informed of the progress of the performance of the services and the Consultant shall refer any decisions which must be made by City to the Contract Officer. Unless otherwise specified herein, any approval of City required hereunder shall mean the approval of the Contract Officer. The Contract Officer shall have authority, if specified in writing by the City Manager, to sign all documents on behalf of the City required hereunder to carry out the terms of this Agreement.

4.4 <u>Independent Consultant.</u>

Neither the City nor any of its employees shall have any control over the manner, mode or means by which Consultant, its agents or employees, perform the services required herein, except as otherwise set forth herein. City shall have no voice in the selection, discharge, supervision or

01007,0001/544964.15

control of Consultant's employees, servants, representatives or agents, or in fixing their number, compensation or hours of service. Consultant shall perform all services required herein as an independent contractor of City and shall remain at all times as to City a wholly independent contractor with only such obligations as are consistent with that role. Consultant shall not at any time or in any manner represent that it or any of its agents or employees are agents or employees of City. City shall not in any way or for any purpose become or be deemed to be a partner of Consultant in its business or otherwise or a joint venturer or a member of any joint enterprise with Consultant.

4.5 <u>Prohibition Against Subcontracting or Assignment.</u>

The experience, knowledge, capability and reputation of Consultant, its principals and employees were a substantial inducement for the City to enter into this Agreement. Therefore, Consultant shall not contract with any other entity to perform in whole or in part the services required hereunder without the express written approval of the City. In addition, neither this Agreement nor any interest herein may be transferred, assigned, conveyed, hypothecated or encumbered voluntarily or by operation of law, whether for the benefit of creditors or otherwise, without the prior written approval of City. Transfers restricted hereunder shall include the transfer to any person or group of persons acting in concert of more than twenty five percent (25%) of the present ownership and/or control of Consultant, taking all transfers into account on a cumulative basis. In the event of any such unapproved transfer, including any bankruptcy proceeding, this Agreement shall be void. No approved transfer shall release the Consultant or any surety of Consultant of any liability hereunder without the express consent of City.

ARTICLE 5. INSURANCE AND INDEMNIFICATION

5.1 Insurance Coverages.

The Consultant shall procure and maintain, at its sole cost and expense, in a form and content satisfactory to City, during the entire term of this Agreement including any extension thereof, the following policies of insurance which shall cover all elected and appointed officers, employees and agents of City:

- (a) <u>General Liability Insurance (Occurrence Form CG0001 or equivalent)</u>. A policy of comprehensive general liability insurance written on a per occurrence basis for bodily injury, personal injury and property damage. The policy of insurance shall be in an amount not less than \$1,000,000.00 per occurrence or if a general aggregate limit is used, then the general aggregate limit shall be twice the occurrence limit.
- (b) <u>Worker's Compensation Insurance</u>. A policy of worker's compensation insurance in such amount as will fully comply with the laws of the State of California and which shall indemnify, insure and provide legal defense for the Consultant against any loss, claim or damage arising from any injuries or occupational diseases occurring to any worker employed by or any persons retained by the Consultant in the course of carrying out the work or services contemplated in this Agreement.
- (c) <u>Automotive Insurance (Form CA 0001 (Ed 1/87) including "any auto" and</u> endorsement CA 0025 or equivalent). A policy of comprehensive automobile liability insurance

written on a per occurrence for bodily injury and property damage in an amount not less than \$1,000,000. Said policy shall include coverage for owned, non-owned, leased, hired cars and any automobile.(d) Professional Liability. Professional liability insurance appropriate to the Consultant's profession. This coverage may be written on a "claims made" basis, and must include coverage for contractual liability. The professional liability insurance required by this Agreement must be endorsed to be applicable to claims based upon, arising out of or related to services performed under this Agreement. The insurance must be maintained for at least 5 consecutive years following the completion of Consultant's services or the termination of this Agreement. During this additional 5-year period, Consultant shall annually and upon request of the City submit written evidence of this continuous coverage.

- (e) <u>Subcontractors</u>. Consultant shall include all subcontractors as insureds under its policies or shall furnish separate certificates and certified endorsements for each subcontractor. All coverages for subcontractors shall include all of the requirements stated herein.
- (f) <u>Additional Insurance</u>. Policies of such other insurance, as may be required in the Special Requirements in Exhibit "B".

5.2 General Insurance Requirements.

All of the above policies of insurance shall be primary insurance and shall name the City, its elected and appointed officers, employees and agents as additional insureds and any insurance maintained by City or its officers, employees or agents may apply in excess of, and not contribute with Consultant's insurance. The insurer is deemed hereof to waive all rights of subrogation and contribution it may have against the City, its officers, employees and agents and their respective insurers. Moreover, the insurance policy must specify that where the primary insured does not satisfy the self-insured retention, any additional insured may satisfy the self-insured retention.

All of said policies of insurance shall provide that said insurance may not be amended or cancelled by the insurer or any party hereto without providing thirty (30) days prior written notice by certified mail return receipt requested to the City. In the event any of said policies of insurance are cancelled, the Consultant shall, prior to the cancellation date, submit new evidence of insurance in conformance with Section 5.1 to the Contract Officer.

No work or services under this Agreement shall commence until the Consultant has provided the City with Certificates of Insurance, additional insured endorsement forms or appropriate insurance binders evidencing the above insurance coverages and said Certificates of Insurance or binders are approved by the City. City reserves the right to inspect complete, certified copies of and endorsements to all required insurance policies at any time. Any failure to comply with the reporting or other provisions of the policies including breaches or warranties shall not affect coverage provided to City.

All certificates shall name the City as additional insured (providing the appropriate endorsement) and shall conform to the following "cancellation" notice:

CANCELLATION:

SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATED THEREOF, THE ISSUING COMPANY SHALL MAIL THIRTY (30)-DAY ADVANCE WRITTEN NOTICE TO CERTIFICATE HOLDER NAMED HEREIN.

[to be initialed]		
-	Consultant Initials	

City, its respective elected and appointed officers, directors, officials, employees, agents and volunteers are to be covered as additional insureds as respects: liability arising out of activities Consultant performs; products and completed operations of Consultant; premises owned, occupied or used by Consultant; or any automobiles owned, leased, hired or borrowed by Consultant. The coverage shall contain no special limitations on the scope of protection afforded to City, and their respective elected and appointed officers, officials, employees or volunteers. Consultant's insurance shall apply separately to each insured against whom claim is made or suit is brought, except with respect to the limits of the insurer's liability.

Any deductibles or self-insured retentions must be declared to and approved by City. At the option of City, either the insurer shall reduce or eliminate such deductibles or self-insured retentions as respects City or its respective elected or appointed officers, officials, employees and volunteers or the Consultant shall procure a bond guaranteeing payment of losses and related investigations, claim administration, defense expenses and claims. The Consultant agrees that the requirement to provide insurance shall not be construed as limiting in any way the extent to which the Consultant may be held responsible for the payment of damages to any persons or property resulting from the Consultant's activities or the activities of any person or persons for which the Consultant is otherwise responsible nor shall it limit the Consultant's indemnification liabilities as provided in Section 5.3.

In the event the Consultant subcontracts any portion of the work in compliance with Section 4.5 of this Agreement, the contract between the Consultant and such subcontractor shall require the subcontractor to maintain the same policies of insurance that the Consultant is required to maintain pursuant to Section 5.1, and such certificates and endorsements shall be provided to City.

5.3 Indemnification.

To the full extent permitted by law, Consultant agrees to indemnify, defend and hold harmless the City, its officers, employees and agents ("Indemnified Parties") against, and will hold and save them and each of them harmless from, any and all actions, either judicial, administrative, arbitration or regulatory claims, damages to persons or property, losses, costs, penalties, obligations, errors, omissions or liabilities whether actual or threatened (herein "claims or liabilities") that may be asserted or claimed by any person, firm or entity arising out of or in connection with the negligent performance of the work, operations or activities provided herein of Consultant, its officers, employees, agents, subcontractors, or invitees, or any individual or entity for which Consultant is legally liable ("indemnitors"), or arising from Consultant's or indemnitors' reckless or willful misconduct, or arising from Consultant's or indemnitors' negligent performance

of or failure to perform any term, provision, covenant or condition of this Agreement, and in connection therewith:

- Consultant will defend any action or actions filed in connection with any of (a) said claims or liabilities and will pay all costs and expenses, including legal costs and attorneys' fees incurred in connection therewith:
- Consultant will promptly pay any judgment rendered against the City, its (b) officers, agents or employees for any such claims or liabilities arising out of or in connection with the negligent performance of or failure to perform such work, operations or activities of Consultant hereunder; and Consultant agrees to save and hold the City, its officers, agents, and employees harmless therefrom:
- In the event the City, its officers, agents or employees is made a party to any action or proceeding filed or prosecuted against Consultant for such damages or other claims arising out of or in connection with the negligent performance of or failure to perform the work, operation or activities of Consultant hereunder, Consultant agrees to pay to the City, its officers, agents or employees, any and all costs and expenses incurred by the City, its officers, agents or employees in such action or proceeding, including but not limited to, legal costs and attorneys' fees.

Consultant shall incorporate similar indemnity agreements with its subcontractors and if it fails to do so Consultant shall be fully responsible to indemnify City hereunder therefore, and failure of City to monitor compliance with these provisions shall not be a waiver hereof. This indemnification includes claims or liabilities arising from any negligent or wrongful act, error or omission, or reckless or willful misconduct of Consultant in the performance of professional services hereunder. The provisions of this Section do not apply to claims or liabilities occurring as a result of City's sole negligence or willful acts or omissions, but, to the fullest extent permitted by law, shall apply to claims and liabilities resulting in part from City's negligence, except that design professionals' indemnity hereunder shall be limited to claims and liabilities arising out of the negligence, recklessness or willful misconduct of the design professional. The indemnity obligation shall be binding on successors and assigns of Consultant and shall survive termination of this Agreement.

5.4 Sufficiency of Insurer.

Insurance required by this Agreement shall be satisfactory only if issued by companies qualified to do business in California, rated "A" or better in the most recent edition of Best Rating Guide, The Key Rating Guide or in the Federal Register, and only if they are of a financial category Class VII or better, unless such requirements are waived by the Risk Manager of the City ("Risk Manager") due to unique circumstances. If this Agreement continues for more than 3 years duration, or in the event the risk manager determines that the work or services to be performed under this Agreement creates an increased or decreased risk of loss to the City, the Consultant

-10-01007.0001/544964.15

agrees that the minimum limits of the insurance policies may be changed accordingly upon receipt of written notice from the Risk Manager.

ARTICLE 6. RECORDS, REPORTS, AND RELEASE OF INFORMATION

6.1 Records.

Consultant shall keep, and require subcontractors to keep, such ledgers, books of accounts, invoices, vouchers, canceled checks, reports, studies or other documents relating to the disbursements charged to City and services performed hereunder (the "books and records"), as shall be necessary to perform the services required by this Agreement and enable the Contract Officer to evaluate the performance of such services. Any and all such documents shall be maintained in accordance with generally accepted accounting principles and shall be complete and detailed. The Contract Officer shall have full and free access to such books and records at all times during normal business hours of City, including the right to inspect, copy, audit and make records and transcripts from such records. Such records shall be maintained for a period of three (3) years following completion of the services hereunder, and the City shall have access to such records in the event any audit is required. In the event of dissolution of Consultant's business, custody of the books and records may be given to City, and access shall be provided by Consultant's successor in interest. Notwithstanding the above, the Consultant shall fully cooperate with the City in providing access to the books and records if a public records request is made and disclosure is required by law including but not limited to the California Public Records Act.

6.2 Reports.

Consultant shall periodically prepare and submit to the Contract Officer such reports concerning the performance of the services required by this Agreement as the Contract Officer shall require. Consultant hereby acknowledges that the City is greatly concerned about the cost of work and services to be performed pursuant to this Agreement. For this reason, Consultant agrees that if Consultant becomes aware of any facts, circumstances, techniques, or events that may or will materially increase or decrease the cost of the work or services contemplated herein or, if Consultant is providing design services, the cost of the project being designed, Consultant shall promptly notify the Contract Officer of said fact, circumstance, technique or event and the estimated increased or decreased cost related thereto and, if Consultant is providing design services, the estimated increased or decreased cost estimate for the project being designed.

6.3 Ownership of Documents.

All drawings, specifications, maps, designs, photographs, studies, surveys, data, notes, computer files, reports, records, documents and other materials (the "documents and materials") prepared by Consultant, its employees, subcontractors and agents in the performance of this Agreement shall be the property of City and shall be delivered to City upon request of the Contract Officer or upon the termination of this Agreement, and Consultant shall have no claim for further employment or additional compensation as a result of the exercise by City of its full rights of ownership use, reuse, or assignment of the documents and materials hereunder. Any use, reuse or assignment of such completed documents for other projects and/or use of uncompleted documents without specific written authorization by the Consultant will be at the City's sole risk and without

01007.0001/544964.15 -11-

liability to Consultant, and Consultant's guarantee and warranties shall not extend to such use, reuse or assignment. Consultant may retain copies of such documents for its own use. Consultant shall have the right to use the concepts embodied therein. All subcontractors shall provide for assignment to City of any documents or materials prepared by them, and in the event Consultant fails to secure such assignment, Consultant shall indemnify City for all damages resulting therefrom. Moreover, Consultant with respect to any documents and materials that may qualify as "works made for hire" as defined in 17 U.S.C. § 101, such documents and materials are hereby deemed "works made for hire" for the City.

6.4 <u>Confidentiality and Release of Information.</u>

- (a) All information gained or work product produced by Consultant in performance of this Agreement shall be considered confidential, unless such information is in the public domain or already known to Consultant. Consultant shall not release or disclose any such information or work product to persons or entities other than City without prior written authorization from the Contract Officer.
- (b) Consultant, its officers, employees, agents or subcontractors, shall not, without prior written authorization from the Contract Officer or unless requested by the City Attorney, voluntarily provide documents, declarations, letters of support, testimony at depositions, response to interrogatories or other information concerning the work performed under this Agreement. Response to a subpoena or court order shall not be considered "voluntary" provided Consultant gives City notice of such court order or subpoena.
- (c) If Consultant, or any officer, employee, agent or subcontractor of Consultant, provides any information or work product in violation of this Agreement, then City shall have the right to reimbursement and indemnity from Consultant for any damages, costs and fees, including attorneys fees, caused by or incurred as a result of Consultant's conduct.
- (d) Consultant shall promptly notify City should Consultant, its officers, employees, agents or subcontractors be served with any summons, complaint, subpoena, notice of deposition, request for documents, interrogatories, request for admissions or other discovery request, court order or subpoena from any party regarding this Agreement and the work performed there under. City retains the right, but has no obligation, to represent Consultant or be present at any deposition, hearing or similar proceeding. Consultant agrees to cooperate fully with City and to provide City with the opportunity to review any response to discovery requests provided by Consultant. However, this right to review any such response does not imply or mean the right by City to control, direct, or rewrite said response.

ARTICLE 7. ENFORCEMENT OF AGREEMENT AND TERMINATION

7.1 <u>California Law.</u>

This Agreement shall be interpreted, construed and governed both as to validity and to performance of the parties in accordance with the laws of the State of California. Legal actions concerning any dispute, claim or matter arising out of or in relation to this Agreement shall be instituted in the Superior Court of the County of Los Angeles, State of California, or any other appropriate court in such county, and Consultant covenants and agrees to submit to the personal

01007.0001/544964.15 -12-

jurisdiction of such court in the event of such action. In the event of litigation in a U.S. District Court, venue shall lie exclusively in the Central District of California, in the County of Los Angeles, State of California.

7.2 <u>Disputes; Default.</u>

In the event that Consultant is in default under the terms of this Agreement, the City shall not have any obligation or duty to continue compensating Consultant for any work performed after the date of default. Instead, the City may give notice to Consultant of the default and the reasons for the default. The notice shall include the timeframe in which Consultant may cure the default. This timeframe is presumptively thirty (30) days, but may be extended, though not reduced, if circumstances warrant. During the period of time that Consultant is in default, the City shall hold all invoices and shall, when the default is cured, proceed with payment on the invoices. In the alternative, the City may, in its sole discretion, elect to pay some or all of the outstanding invoices during the period of default. If Consultant does not cure the default, the City may take necessary steps to terminate this Agreement under this Article. Any failure on the part of the City to give notice of the Consultant's default shall not be deemed to result in a waiver of the City's legal rights or any rights arising out of any provision of this Agreement.

7.3 Retention of Funds.

Consultant hereby authorizes City to deduct from any amount payable to Consultant (whether or not arising out of this Agreement) (i) any amounts the payment of which may be in dispute hereunder or which are necessary to compensate City for any losses, costs, liabilities, or damages suffered by City, and (ii) all amounts for which City may be liable to third parties, by reason of Consultant's acts or omissions in performing or failing to perform Consultant's obligation under this Agreement. In the event that any claim is made by a third party, the amount or validity of which is disputed by Consultant, or any indebtedness shall exist which shall appear to be the basis for a claim of lien, City may withhold from any payment due, without liability for interest because of such withholding, an amount sufficient to cover such claim. The failure of City to exercise such right to deduct or to withhold shall not, however, affect the obligations of the Consultant to insure, indemnify, and protect City as elsewhere provided herein.

7.4 Waiver.

Waiver by any party to this Agreement of any term, condition, or covenant of this Agreement shall not constitute a waiver of any other term, condition, or covenant. Waiver by any party of any breach of the provisions of this Agreement shall not constitute a waiver of any other provision or a waiver of any subsequent breach or violation of any provision of this Agreement. Acceptance by City of any work or services by Consultant shall not constitute a waiver of any of the provisions of this Agreement. No delay or omission in the exercise of any right or remedy by a non-defaulting party on any default shall impair such right or remedy or be construed as a waiver. Any waiver by either party of any default must be in writing and shall not be a waiver of any other default concerning the same or any other provision of this Agreement.

01007.0001/544964.15 -13-

7.5 Rights and Remedies are Cumulative.

Except with respect to rights and remedies expressly declared to be exclusive in this Agreement, the rights and remedies of the parties are cumulative and the exercise by either party of one or more of such rights or remedies shall not preclude the exercise by it, at the same or different times, of any other rights or remedies for the same default or any other default by the other party.

7.6 Legal Action.

In addition to any other rights or remedies, either party may take legal action, in law or in equity, to cure, correct or remedy any default, to recover damages for any default, to compel specific performance of this Agreement, to obtain declaratory or injunctive relief, or to obtain any other remedy consistent with the purposes of this Agreement. Notwithstanding any contrary provision herein, Consultant shall file a statutory claim pursuant to Government Code Sections 905 et. seq. and 910 et. seq., in order to pursue a legal action under this Agreement.

7.7 <u>Termination Prior to Expiration of Term.</u>

This Section shall govern any termination of this Contract except as specifically provided in the following Section for termination for cause. The City reserves the right to terminate this Contract at any time, with or without cause, upon thirty (30) days' written notice to Consultant, except that where termination is due to the fault of the Consultant, the period of notice may be such shorter time as may be determined by the Contract Officer. In addition, the Consultant reserves the right to terminate this Contract at any time, with or without cause, upon sixty (60) days' written notice to City, except that where termination is due to the fault of the City, the period of notice may be such shorter time as the Consultant may determine. Upon receipt of any notice of termination, Consultant shall immediately cease all services hereunder except such as may be specifically approved by the Contract Officer. Except where the Consultant has initiated termination, the Consultant shall be entitled to compensation for all services rendered prior to the effective date of the notice of termination and for any services authorized by the Contract Officer thereafter in accordance with the Schedule of Compensation or such as may be approved by the Contract Officer, except as provided in Section 7.3. In the event the Consultant has initiated termination, the Consultant shall be entitled to compensation only for the reasonable value of the work product actually produced hereunder. In the event of termination without cause pursuant to this Section, the terminating party need not provide the non-terminating party with the opportunity to cure pursuant to Section 7.2.

7.8 Termination for Default of Consultant.

If termination is due to the failure of the Consultant to fulfill its obligations under this Agreement, City may, after compliance with the provisions of Section 7.2, take over the work and prosecute the same to completion by contract or otherwise, and the Consultant shall be liable to the extent that the total cost for completion of the services required hereunder exceeds the compensation herein stipulated (provided that the City shall use reasonable efforts to mitigate such

-14-01007.0001/544964.15

damages), and City may withhold any payments to the Consultant for the purpose of set-off or partial payment of the amounts owed the City as previously stated.

7.9 <u>Attorneys' Fees.</u>

If either party to this Agreement is required to initiate or defend or made a party to any action or proceeding in any way connected with this Agreement, the prevailing party in such action or proceeding, in addition to any other relief which may be granted, whether legal or equitable, shall be entitled to reasonable attorney's fees. Attorney's fees shall include attorney's fees on any appeal, and in addition a party entitled to attorney's fees shall be entitled to all other reasonable costs for investigating such action, taking depositions and discovery and all other necessary costs the court allows which are incurred in such litigation. All such fees shall be deemed to have accrued on commencement of such action and shall be enforceable whether or not such action is prosecuted to judgment.

ARTICLE 8. CITY OFFICERS AND EMPLOYEES: NON-DISCRIMINATION

8.1 Non-liability of City Officers and Employees.

No officer or employee of the City shall be personally liable to the Consultant, or any successor in interest, in the event of any default or breach by the City or for any amount which may become due to the Consultant or to its successor, or for breach of any obligation of the terms of this Agreement.

8.2 Conflict of Interest.

Consultant covenants that neither it, nor any officer or principal of its firm, has or shall acquire any interest, directly or indirectly, which would conflict in any manner with the interests of City or which would in any way hinder Consultant's performance of services under this Agreement. Consultant further covenants that in the performance of this Agreement, no person having any such interest shall be employed by it as an officer, employee, agent or subcontractor without the express written consent of the Contract Officer. Consultant agrees to at all times avoid conflicts of interest or the appearance of any conflicts of interest with the interests of City in the performance of this Agreement.

No officer or employee of the City shall have any financial interest, direct or indirect, in this Agreement nor shall any such officer or employee participate in any decision relating to the Agreement which affects her/his financial interest or the financial interest of any corporation, partnership or association in which (s)he is, directly or indirectly, interested, in violation of any State statute or regulation. The Consultant warrants that it has not paid or given and will not pay or give any third party any money or other consideration for obtaining this Agreement.

8.3 Covenant Against Discrimination.

Consultant covenants that, by and for itself, its heirs, executors, assigns, and all persons claiming under or through them, that there shall be no discrimination against or segregation of, any person or group of persons on account of race, color, creed, religion, sex, gender, sexual orientation, marital status, national origin, ancestry or other protected class in the performance of

01007.0001/544964.15 -15-

this Agreement. Consultant shall take affirmative action to insure that applicants are employed and that employees are treated during employment without regard to their race, color, creed, religion, sex, gender, sexual orientation, marital status, national origin, ancestry or other protected class.

8.4 <u>Unauthorized Aliens.</u>

Consultant hereby promises and agrees to comply with all of the provisions of the Federal Immigration and Nationality Act, 8 U.S.C.A. §§ 1101, *et seq.*, as amended, and in connection therewith, shall not employ unauthorized aliens as defined therein. Should Consultant so employ such unauthorized aliens for the performance of work and/or services covered by this Agreement, and should any liability or sanctions be imposed against City for such use of unauthorized aliens, Consultant hereby agrees to and shall reimburse City for the cost of all such liabilities or sanctions imposed, together with any and all costs, including attorneys' fees, incurred by City.

ARTICLE 9. MISCELLANEOUS PROVISIONS

9.1 Notices.

Any notice, demand, request, document, consent, approval, or communication either party desires or is required to give to the other party or any other person shall be in writing and either served personally or sent by prepaid, first-class mail, in the case of the City, to the City Manager and to the attention of the Contract Officer (with her/his name and City title), City of Carson, 701 East Carson, Carson, California 90745 and in the case of the Consultant, to the person(s) at the address designated on the execution page of this Agreement. Either party may change its address by notifying the other party of the change of address in writing. Notice shall be deemed communicated at the time personally delivered or in seventy-two (72) hours from the time of mailing if mailed as provided in this Section.

9.2 Interpretation.

The terms of this Agreement shall be construed in accordance with the meaning of the language used and shall not be construed for or against either party by reason of the authorship of this Agreement or any other rule of construction which might otherwise apply.

9.3 Counterparts.

This Agreement may be executed in counterparts, each of which shall be deemed to be an original, and such counterparts shall constitute one and the same instrument.

9.4 Integration; Amendment.

This Agreement including the attachments hereto is the entire, complete and exclusive expression of the understanding of the parties. It is understood that there are no oral agreements between the parties hereto affecting this Agreement and this Agreement supersedes and cancels any and all previous negotiations, arrangements, agreements and understandings, if any, between the parties, and none shall be used to interpret this Agreement. No amendment to or modification of this Agreement shall be valid unless made in writing and approved by the Consultant and by

01007.0001/544964.15 -16-

the City Council. The parties agree that this requirement for written modifications cannot be waived and that any attempted waiver shall be void.

9.5 <u>Severability.</u>

In the event that any one or more of the phrases, sentences, clauses, paragraphs, or sections contained in this Agreement shall be declared invalid or unenforceable by a valid judgment or decree of a court of competent jurisdiction, such invalidity or unenforceability shall not affect any of the remaining phrases, sentences, clauses, paragraphs, or sections of this Agreement which are hereby declared as severable and shall be interpreted to carry out the intent of the parties hereunder unless the invalid provision is so material that its invalidity deprives either party of the basic benefit of their bargain or renders this Agreement meaningless.

9.6 Warranty & Representation of Non-Collusion.

No official, officer, or employee of City has any financial interest, direct or indirect, in this Agreement, nor shall any official, officer, or employee of City participate in any decision relating to this Agreement which may affect his/her financial interest or the financial interest of any corporation, partnership, or association in which (s)he is directly or indirectly interested, or in violation of any corporation, partnership, or association in which (s)he is directly or indirectly interested, or in violation of any State or municipal statute or regulation. The determination of "financial interest" shall be consistent with State law and shall not include interests found to be "remote" or "non-interests" pursuant to Government Code Sections 1091 or 1091.5. Consultant warrants and represents that it has not paid or given, and will not pay or give, to any third party including, but not limited to, any City official, officer, or employee, any money, consideration, or other thing of value as a result or consequence of obtaining or being awarded any agreement. Consultant further warrants and represents that (s)he/it has not engaged in any act(s), omission(s), or other conduct or collusion that would result in the payment of any money, consideration, or other thing of value to any third party including, but not limited to, any City official, officer, or employee, as a result of consequence of obtaining or being awarded any agreement. Consultant is aware of and understands that any such act(s), omission(s) or other conduct resulting in such payment of money, consideration, or other thing of value will render this Agreement void and of no force or effect.

Consultant's Authorized Initials _____

9.7 <u>Corporate Authority.</u>

The persons executing this Agreement on behalf of the parties hereto warrant that (i) such party is duly organized and existing, (ii) they are duly authorized to execute and deliver this Agreement on behalf of said party, (iii) by so executing this Agreement, such party is formally bound to the provisions of this Agreement, and (iv) the entering into this Agreement does not violate any provision of any other Agreement to which said party is bound. This Agreement shall be binding upon the heirs, executors, administrators, successors and assigns of the parties.

[SIGNATURES ON FOLLOWING PAGE]

01007.0001/544964.15 -17-

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on the date and year first-above written.

	CITY:
	CITY OF CARSON, a municipal corporation
ATTEST:	Lula Davis-Holmes, Mayor
Donesia Gause-Aldana, City Clerk	
APPROVED AS TO FORM: ALESHIRE & WYNDER, LLP	
Sunny K. Soltani, City Attorney [brj and rjl]	
	CONSULTANT:
	ACTIVE NETWORK, LLC, a Delaware limited liability company
	By: Name: Andrea Facini Title: President Address: 717 N. Harwood, Suite 2500 Dallas, TX 75201

Two corporate officer signatures required when Consultant is a corporation, with one signature required from each of the following groups: 1) Chairman of the Board, President or any Vice President; and 2) Secretary, any Assistant Secretary, Chief Financial Officer or any Assistant Treasurer. CONSULTANT'S SIGNATURES SHALL BE DULY NOTARIZED, AND APPROPRIATE ATTESTATIONS SHALL BE INCLUDED AS MAY BE REQUIRED BY THE BYLAWS, ARTICLES OF INCORPORATION, OR OTHER RULES OR REGULATIONS APPLICABLE TO CONSULTANT'S BUSINESS ENTITY.

01007.0001/544964.15 -18-

CALIFORNIA ALL-PURPOSE ACKNOWLEDGMENT

A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy or validity of that document.

STATE OF CALIFORNIA		
COUNTY OF LOS ANGELES		
On, 2021 before me,, personally appeared, proved to me on the basis of satisfactory evidence to be the person(s) whose names(s) is/are subscribed to the within instrument and acknowledged to me that he/she/they executed the same in his/her/their authorized capacity(ies), and that by his/her/their signature(s) on the instrument the person(s), or the entity upon behalf of which the person(s) acted, executed the instrument.		
I certify under PENALTY OF PERJURY under the laws of th and correct.	e State of California that the foregoing paragraph is true	
WITNESS my hand and official seal.		
Signature:		
OPTIONAL Though the data below is not required by law, it may prove valuable to persons relying on the document and could prevent fraudulent reattachment of this form. CAPACITY CLAIMED BY SIGNER DESCRIPTION OF ATTACHED DOCUMENT		
☐ INDIVIDUAL ☐ CORPORATE OFFICER	DESCRIPTION OF HEALTH DOCUMENT	
TITLE(S) PARTNER(S) LIMITED GENERAL ATTORNEY-IN-FACT	TITLE OR TYPE OF DOCUMENT	
TRUSTEE(S) GUARDIAN/CONSERVATOR OTHER	NUMBER OF PAGES	
SIGNER IS REPRESENTING: (NAME OF PERSON(S) OR ENTITY(IES)) DATE OF DOCUMENT		
	SIGNER(S) OTHER THAN NAMED AROVE	

CALIFORNIA ALL-PURPOSE ACKNOWLEDGMENT

A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy or validity of that document.

STATE OF CALIFORNIA		
COUNTY OF LOS ANGELES		
On, 2021 before me,, personally appeared, proved to me on the basis of satisfactory evidence to be the person(s) whose names(s) is/are subscribed to the within instrument and acknowledged to me that he/she/they executed the same in his/her/their authorized capacity(ies), and that by his/her/their signature(s) on the instrument the person(s), or the entity upon behalf of which the person(s) acted, executed the instrument.		
I certify under PENALTY OF PERJURY under the laws of t	he State of California that the foregoing paragraph is true	
WITNESS my hand and official seal.		
Signature:		
OPTIONAL Though the data below is not required by law, it may prove valuable to persons relying on the document and could prevent fraudulent reattachment of this form.		
CAPACITY CLAIMED BY SIGNER INDIVIDUAL CORPORATE OFFICER	DESCRIPTION OF ATTACHED DOCUMENT	
TITLE(S) PARTNER(S) LIMITED GENERAL ATTORNEY-IN-FACT	TITLE OR TYPE OF DOCUMENT	
TRUSTEE(S) GUARDIAN/CONSERVATOR OTHER	NUMBER OF PAGES	
SIGNER IS REPRESENTING: (NAME OF PERSON(S) OR ENTITY(IES)) DATE OF DOCUMENT		
	SIGNER(S) OTHER THAN NAMED ABOVE	

EXHIBIT "A" SCOPE OF SERVICES

I. Consultant will perform the following Services:

Provide recreation management software that facilitates recreation program registration, indoor and outdoor facility reservations, fitness and boxing program memberships, citywide youth sports league management, point of sale (POS), marketing, and reporting, and install the necessary hardware (equipment) and deliver software implementation services. This includes provision to the City with a subscription to "ACTIVE Net," Consultant's fully hosted, web-based "Software as a System" ("SaaS") solution, for use by City on an unlimited number of workstations and by an unlimited number of concurrent users, giving the City ondemand access to its data from any computer with an internet connection. ACTIVE Net will also include an *ACTIVE.com* portal that will allow community members to view and book reservations for use of City facilities or participation in City programs, events, or memberships (including classes and camps) online. Consultant will provide ACTIVE Net to City in accordance with its "Service Package Advanced 7," which includes Professional Services such as:

- 1. Remote business process review (meeting to assess the IT aspects of the implementation);
- 2. Remote functionality review & data collection preparation;
- 3. Remote data collection review:
- 4. Remote data entry (inventory and policy controls);
- 5. Remote user testing
- 6. LMS training;
- 7. Remote supplemental training;
- 8. Remote Go Live preparation;
- 9. Remote hardware configuration; and
- 10. Remote system optimization training.

Senior consulting services will be delivered by Consultant's Enterprise Consulting team, representing its most seasoned agents.

Consultant shall perform the following in connection with the Service Package Advanced 7:

A. PROJECT PLANNING:

Project Launch/Scheduling

Objectives for the Project Launch/Scheduling stage are:

- Gain an understanding of required project roles
- Gain understanding of the project life cycle
- Establish a tentative project schedule
- Reserve Active and Customer resources

• Prepare for the Business Process Review

Inputs for the Project Launch/Scheduling stage are:

•	Active Resource Availability	ACTIVE
•	Customer Project Resource Availability	Customer PM
•	Project Planning Guide	ACTIVE
•	Data Assessment Sheet (blank)	ACTIVE
•	Deliver Accounting Manual	ACTIVE
•	ACTIVENet Look and Feel Guide	ACTIVE
•	Technical Statement of Works (SOWs)	ACTIVE

Tasks for the Project Launch/Scheduling stage are:

- Review ACTIVE project roles and responsibilities
- Review City's required project roles and responsibilities
- Review ACTIVENet's project lifecycle
- Review project schedule options
- Review Basecamp use
- Distribute the Data Assessment Sheets and other documents

Deliverables for the Project Launch/Scheduling stage are:

•	Project schedule	ACTIVE
•	Active Project team assignments	ACTIVE
•	Customer project team assignments	Customer PM
•	Complete Data Assessment Sheet	Customer SME
•	Review Accounting Manual	Customer
	Accounting Team	
•	Review Technical SOWs	Customer Technical Team
•	Review Recommended Settings document	Customer Technical Team
•	Booking the Business Review meeting room	Customer PM
•	Setup Basecamp instance and provide login credentials	ACTIVE

Business Process Review

Objectives for the Business Review stage are:

- Understand City's project goals
- Understand City's business policies and procedures
- Understand City's product and service offerings and establish a data collection strategy
- Understand City's IT infrastructure
- Understand City's accounting needs
- Identify project risks
- Finalize project schedule and go live date

Inputs for the Business Review stage are:

•	Completed data assessment sheet	Customer SME
•	Reviewed Accounting Manual	Customer
	Accounting Team	
•	General Data Collection Sheet (Blank)	ACTIVE
•	Provide images and waivers to Basecamp	Customer SME

Tasks for the Business Review stage are:

• Review project goals

- Review the Data Assessment Sheet
- Review General Data Collection Sheet so that City knows how to fill it out
- Discuss the other Data Collection Sheets at a high level
- Discuss accounting needs
- Review technical system requirements
- Discuss peripherals being used by City
- Create a communication plan (for project team, City stakeholders, staff and customers)
- Maintain a Project Risk Log
- Review Active Learning Management System (LMS) and plan out courses for next sessions

Deliverables for the Business Review stage are:

•	Data Assessment Sheet (populated)	Customer
	SME/ACTIVE	
•	Data Collection Sheets (blank)	ACTIVE
•	Project Risk Log	Customer PM
•	Project Agenda for the remaining stages	ACTIVE
•	Basecamp dates and tasks updated	ACTIVE
•	LMS link and logins	ACTIVE

B. IMPLEMENTATION:

Data Collection

Objectives for the Data Collection stage are:

- Capture City's inventory of all product and services in a format for use during data entry.
- Review/Overview of the ACTIVENet modules to be used

Inputs for the Data Collection stage are:

• Provide maps to Basecamp

Customer SME

• Data Collection Sheets (completed)

Customer SME

Tasks for the Data Collection stage are:

- Populate the Data Assessment Sheets with inventory of all product and services.
- Complete assigned LMS courses prior to the sessions
- Determine who will enter the data (Active and/or the City)
- Demonstrate standard workflows of ACTIVENet modules

Deliverables for the Data Collection stage are:

• Data Collection Sheets (Populated)

Customer SME

Data Review

Objectives for the Data Review stage are:

- Ensure the Data Collection Sheets are complete and accurate
- Initial introduction to the ACTIVENet application modules

Inputs for the Data Review stage are:

Completed data assessment sheet

Customer SME

Tasks for the Data Review stage are:

- Review the Data Collection Sheets for completeness
- Perform validation checks on data collection sheets

Assign Learning Management System courses to staff for testing phase

Deliverables for the Data Review stage are:

• Data Collection Sheets (Populated)

Customer SME

Data Entry

- **Objectives** for the Data Entry stage are:
- Configure and populate ACTIVENet database in accordance with City's needs

Inputs for the Data Entry stage are:

• Competed data assessment sheet

Customer SME

Tasks for the Data Entry stage are:

- Configure ACTIVENet in conjunction with the Data Assessment Sheet
- Populate ACTIVENet with product and services from the Data Collection Sheets. This is non-customer/transaction data.
- Audit ACTIVENet to ensure alignment with Data Collection Sheets

Deliverables for the Data Entry stage are:

• ACTIVENet database (Configured and populated)

ACTIVE

User Testing

Objectives for the User Testing stage are:

- Test the functional integrity of the system configuration against business cases
- Update configuration as needed

Inputs for the User Testing stage are:

• User testing templates

ACTIVE

• Test cases and workflows to review

Customer SME

Tasks for the User Testing stage are:

- Review front-desk transactions for each functional area in ACTIVENet
- Review online transactions for each functional area in ACTIVENet
- Review reports for each functional area in ACTIVENet
- Review communication tools for each functional area in ACTIVENet
- Determine mitigation strategies for each functional gap
- Update configuration in accordance with testing results
- End to end test of credit card processing and bank reconciliation

Deliverables from the User Testing stage are:

ACTIVENet (tested)
 Active/Customer

SME

• Configuration is updated as needed Active and

Customer SME

• Detailed review of ACTIVENet modules Customer SME

• Updated test cases Customer SME

Training (LMS Train the Trainer)

Objectives for the Training stage are:

- Introduce organization trainers to the Active online LMS training tool
- Empower the project team to understand best practices for conducting end-user training

- Empower the project team with a strategy and tools to conduct end-user training.
- Create and update training material for the end users
- Create an end user training plan

Inputs for the Training stage are:

Training Manual template and quick reference guides
 Provide list of Organization trainers
 Training room booked and prepared
 Customer PM
 Customer PM

Tasks for the Training stage are:

- Provide base training materials
- Trainers have completed the assigned LMS courses prior to the sessions
- Modify training manual in accordance with system usage
- Provide supplemental training session on workflows not covered in the LMS
- Develop training plan for delivering end-user staff training
- Secure resources to execute training plan

Deliverables for the Training stage are:

Customized Training Manual
 Training Plan
 Customer Trainers
 Customer PM

C. DEPLOYMENT:

Workstation Readiness

Objectives for the Workstation Readiness stage are:

- Ensure all workstations have access to ACTIVENet
- Ensure all workstations designated to operate ACTIVENet are configured with the appropriate 3rd party software
- Ensure all workstations designated to operate ACTIVENet are configured with the appropriate hardware

Inputs for the Workstation Readiness stage are:

•	Existing Peripheral Hardware List	Customer PM
•	Additional Peripheral Hardware (if required)	ACTIVE
•	System Requirements	ACTIVE
•	Hardware Configuration Guides	ACTIVE

Tasks for the Workstation Readiness stage are:

- Install, configure and test third-party software required by ACTIVENet
- Install, configure and test hardware required by ACTIVENet
- Bookmark ACTIVENet sites on workstations

Deliverables for the Workstation Readiness stage are:

- Configure and test single workstation with peripherals (ACTIVE and Customer Desktop)
- Configure and test remaining workstations and peripherals Customer Desktop

End User Training

Objectives for the End User Training stage are:

- Train system users on software operations
- Communicate software escalation procedures

Inputs for the End User Training stage are:

End-user training guide
 Training plan
 Training rooms are booked and prepared
 Customer Trainers
 Customer PM

Tasks for the End User Training stage are:

• Execute training plan

Deliverables for the End User Training stage are:

All Staff trained
 Customer Trainers

D. GO-LIVE:

Go-Live Preparation

Consultant will conduct a meeting prior to go-live to confirm all tasks are complete and the system is ready to be the City's true system of record.

Objectives for the Go-Live Preparation stage are:

- Ensure staff are competent in system use
- Ensure workstations are ready for system use
- Ensure data conversion is complete
- Ensure project team understands how to escalate issues to ACTIVE
- Ensure risks are captured and mitigation strategies are in place

Inputs for the Go-Live Preparation stage are:

Prior stages are complete
 ACTIVE Support handbook
 Go-Live Checklist

Customer PM
ACTIVE
ACTIVE

Tasks for the Go-Live Preparation stage are:

- Review and validate staff training
- Review and validate workstation readiness
- Review and validate data conversion completeness
- Review Active Support policy and communication channels
- Handoff call with ACTIVE support team.
- Complete Go-Live checklist

Deliverables for the Go-Live Preparation stage are:

Go Live checklist
 Customer Support plan
 Go Live plan
 Customer PM
 Customer PM

Go-Live

Go-Live is the day that ACTIVENet becomes the system of record for the City.

Objectives for the Go-Live stage are:

• Help with any issues or configuration questions

Inputs for the Go-Live stage are:

• Go live plan Customer PM

Tasks for the Go-Live stage are:

• All prior stages are complete

Deliverables for the Go-Live stage are:

• Online payments are enabled

Customer PM

• End users processing in ACTIVENet

Customer PM

Post-Go-Live System Optimization

Consultant will conduct a meeting 2 to 3 weeks after go-live to review any questions or follow up configuration required.

Objectives for Post Go-Live Optimization stage are:

- Optimize usage of ACTIVENet to perform front desk transactions and manage operations
- Optimize usage of ACTIVENet to provide City an online store

Inputs for the Post Go-Live Optimization stage are:

• List of questions, issues and topics

Customer PM

Tasks for the Post Go-Live Optimization stage are:

• Session with the consultant

E. TRANSPORATION OF PRODUCTS:

Primary transportation of all ACTIVE Net-related products and equipment to necessary locations for installation and City use;

F. MAINTENANCE:

Maintenance Services related to ensuring full ongoing availability and functionality of ACTIVE Net, to be provided in accordance with Consultant's "Advanced Support Package," which includes, without limitation:

- 1. Responding to an unlimited number of support requests to Consultant's customer support analysts accessible via phone, messaging, or email;
 - (i) 24/7/365 emergency phone support;
 - (ii) live phone support during Consultant's business hours (subject to change);
 - (iii) target email response time of 24 hours;
- 2. 24/7/365 access to online customer portal and self-service support via searchable databases, user guides & online forums;
- 3. 24/7/365 web/email case submission;
- 4. A 15-user learning management system;
- 5. Rush event support;

G. PAYMENT REMITTANCE TECHNICAL SERVICES:

ACTIVE Net ACH (Automated Clearing House) Payment Remittance Technical Services ("Daily" - Every 72 Hours); ACTIVE Net ACH Payment Remittance Technical Services (Every 2 weeks);

- **H.** Provide ACTIVE Net "Train the Trainer" training to the following City staff members at the following competency levels, unless otherwise requested by the Contract Officer (and approved by the City Council if additional cost exceeding Contract Officer authority pursuant to Section 1.8 is involved):
 - 1. Two recreation staff members Administration;
 - 2. Two recreation staff members Registration;
 - 3. One Finance staff member All Financial Aspects;
 - 4. One Information Technology stff member tech support.

Training shall be provided in accordance with the content and timeline set forth in Consultant's proposal, unless otherwise set forth herein or requested by the Contract Officer.

I. DETAILED BREAKDOWN:

In connection with Consultant's provision of Service Package Advanced 7 and the foregoing, Consultant shall provide City with the following services and products, identified as "Tasks":

TASKS

HARDWARE - YEAR 1 ONLY:

- 1. PAX 300 Pinpad Device
- 2. Topaz Digital Signature Pad (may not be necessary)
- 3. APG Cash Drawer (may not be necessary)
- 4. Barcode and Smartphone Scanner (may not be necessary)

SOFTWARE AND LICENSES -- YEARS 1 THROUGH 3:

- 5. Annual Subscription (includes: Support, Unlimited Licenses, access to solution, Hosting costs, Upgrades, Database Backups, etc.)
- 6. ACTIVENet Connect (may not be necessary)
- 7. ACTIVENet Captivate (may not be necessary)

PROFESSIONAL SERVICES -- YEAR 1 ONLY:

- 8. Implementation Advanced 7 Service Package (Implementation, Configuration, Training)
- 9. Financial Export

- 10. Brochure Export
- 11. GIS Export
- II. In addition to the requirements of Section 6.2, during performance of the Services, Consultant will keep the City apprised of the status of performance by delivering the following status reports:

Reports containing all registration information, full calculations with detailed support, and all background information necessary to verify and substantiate such calculations, and full statements including rates, quantities, and totals, for all Fees collected by Consultant from End Users and fees disbursed to City by Consultant pursuant to this Agreement, shall be available to City within the ACTIVE Net product through the primary reporting modules for the Term of this Agreement.

III. All data obtained by Consultant shall be encrypted in accordance with industry standard while in Consultant's possession. Notwithstanding anything else to the contrary, within a reasonable time upon expiration or sooner termination of this Agreement but no more than thirty (30) days thereof, Consultant will deliver to City a file containing all City data for download in a commercially-reasonable standard format, such as comma separated value (.csv) or extendible markup language (.xml), along with attachments in their native format as stored by Consultant. Consultant shall thereafter delete all City data in Consultant's systems or otherwise in Consultant's possession or under Consultant's control. If requested by City, Consultant will certify that it has returned or destroyed all copies of the applicable City data using commercially reasonable measures and acknowledges that its rights to use the same are relinquished. There shall be a \$3,000 charge to City for this data export service.

EXHIBIT "B" SPECIAL REQUIREMENTS

(Superseding Contract Boilerplate)

The Agreement is hereby amended as follows (deletions shown in strikethrough and additions shown in *bold italics*):

I. Section 1.5, Familiarity With Work, is hereby amended to read in its entirety as follows:

"1.5 <u>Familiarity with Work.</u> By executing this Agreement, Consultant warrants agrees that Consultant (i) has thoroughly investigated and considered the scope of services to be performed, (ii) has carefully considered how the services should be performed, and (iii) fully understands the facilities, difficulties and restrictions attending performance of the services under this Agreement. If the services involve work upon any site, Consultant warrants that Consultant has or will investigate the site and is or will be fully acquainted with the conditions there existing, prior to commencement of services hereunder. Should the Consultant discover any latent or unknown conditions, which will materially affect the performance of the services hereunder, Consultant shall immediately inform the City of such fact and shall not proceed except at Consultant's risk until written instructions are received from the Contract Officer make commercially reasonable efforts to remedy the defect as swiftly as possible."

II. Section 1.6, Care of Work, is hereby amended to read in its entirety as follows:

"1.6 <u>Care of Work.</u> The Consultant shall adopt reasonable methods during the life of the Agreement to furnish continuous protection to the work, and the equipment, materials, papers, documents, plans, studies and/or other components thereof to prevent losses or damages, and, *pursuant to the Indemnification and Limitation of Liability provisions of this Agreement*, shall be responsible for all such damages, to persons or property, until acceptance of the work by City, except such losses or damages as may be caused by City's own negligence the City."

III. Section 1.9, Special Requirements, is hereby amended to read in its entirety as follows:

"1.9 Special Requirements. Additional terms and conditions of this Agreement, if any, which are made a part hereof are set forth in the "Special Requirements" attached hereto as Exhibit "B" and incorporated herein by this reference. In the event of a conflict between the provisions of Exhibit "B" and any other provisions of this Agreement, the provisions of Exhibit "B" shall govern. No additional terms and conditions will be added to this Agreement or Exhibit "B" unless mutually agreed upon by both parties in writing."

IV. Section 2.1, Contract Sum, is hereby renamed and amended to read in its entirety as follows:

"2.1 Contract Sum. Payment terms are specified in the Schedule of Compensation (Exhibit "C"). Unless otherwise specified in the Schedule of Compensation, all amounts owed by City are due from City within 30 days from the date of the applicable invoice. Subject to any limitations set forth in this Agreement, City agrees to pay Consultant the amounts specified in the "Schedule of Compensation" attached hereto as Exhibit "C" and incorporated herein by this reference. The total compensation, including reimbursement for actual expenses, shall not exceed TWO HUNDRED FIFTY THOUSAND DOLLARS (\$250,000) for the initial term of this Agreement, in accordance with the following: (1) ONE HUNDRED THIRTY THOUSAND DOLLARS (\$130,000) for the first year of the Term, (2) SIXTY THOUSAND DOLLARS (\$60,000) for the second year of the Term, and (3) SIXTY THOUSAND DOLLARS (\$60,000) for the third year of the Term (collectively, the "Contract Sum"), unless additional compensation is approved pursuant to Section 1.8. In the event City exercises any or all of its three options to extend the term of this Agreement as set forth in Section 3.4, the compensation for the first year of the first option period shall not exceed five percent (5%) more than \$60,000, and the compensation for each sucessive year thereafter for any option period shall increase by no more than five percent (5%) of the amount applicable for the immediately preceding year. Any annual increase in compensation during any option period shall be determined by Consultant in accordance with any reasonable cost of living adjustments.

Unless otherwise set forth in the Schedule of Compensation, Consultant will charge Fees to individuals who register for the Events or purchase goods or services online, and will process and collect such Fees as a merchant of record according to the card networks. On a bi-weekly basis, Consultant will pay City sums due to City based on the total Event registration revenues collected, net of any deductions provided herein. The applicable currency will be U.S. Dollars. In no event shall City pay nor be responsible for payment of any Fees."

V. Section 2.2, Method of Compensation, is hereby deleted in its entirety and replaced with the following:

"2.2. Method of Compensation.

(a) <u>Costs.</u> The method of compensation shall be as follows: City will be invoiced on a quarterly basis commencing upon the date of the first live operational use of the SaaS for the Event(s) ("Go-Live Date"), with costs being invoiced in increments of 25% of the annual amount, commencing upon the Go-Live Date.

If there are any overdue amounts owed by City, or there are returned charges or items, including those resulting from any error or complaint related to an Event, Consultant has the right to charge fees owed to Consultant by City by issuing an

invoice, or by offsetting the deficiency from any account balance City maintains with Consultant or any payment Consultant owes City.

- (b) Fees. Consultant will charge Fees (as defined in Section III of Exhibit "C") to individuals who register for Events or purchase goods or services using the SaaS pursuant to this Agreement, and will process and collect such fees as a merchant of record according to the card networks in accordance with Section III of Exhibit "C." Consultant will pay City sums due to City based on the total fees collected on a bi-weekly basis in accordance with Section 7.3 and Section III of Exhibit "C." All Fees are in consideration of the SaaS and Services that Consultant provides. Consultant and City acknowledge that certain credit card network rules and laws prohibit imposing a surcharge that is based on the type of payment method used (e.g., having a different fee for the use of a credit card vs. debit card), and therefore, each agrees not to impose such a surcharge on any End User.
- It is City's responsibility to notify End Users of City's refund policy. City must ensure that City's refund policies are consistent with the Agreement. City agrees that all fees for a given Event are earned by City only following either the conclusion or delivery of the applicable Event (as applicable) and all amounts ultimately due to City will be net of all reversals, refunds, disputed charges, chargebacks, and other deductions, whether due to customer complaints, allegations of fraud, discrepancies related to the applicable Event, or otherwise, as authorized in this Agreement. No payments will be made to City with respect to any Event that is cancelled by City. If payments have already been made by Consultant to City for a cancelled Event or if Consultant is required by law to pay a refund to or honor a chargeback request from an End User, Consultant may issue an invoice or offset an equivalent amount from City's account or payment owed by Consultant to City and return the value to the End User, and if sufficient funds are not available, City must reimburse Consultant on demand. Consultant will notify City of the amount and date of such offset, and the reason for such offset provided that it is lawful to do so. Except as otherwise provided in this paragraph, refunds to End Users shall be made only by City in accordance with City's refund policy."

VI. A new Section 2.3, Taxes, is hereby added to the Agreement, to read in its entirety as follows:

"2.3 Taxes. Consultant's rates, prices and fees set forth in Exhibit "C" may (where indicated) or may not include Taxes. City is responsible for and agrees to pay any and all Taxes applicable to City. City is solely responsible for determining which, if any, Taxes apply to City's use of the Products and for collecting, remitting, and reporting the correct amounts of all such Taxes to the applicable governmental authorities, even if Consultant provides City with tools that assist City in doing so. In the event that a governmental authority requires Consultant to pay any Taxes attributable to City's use of the Products, City agrees to defend, indemnify, and hold Consultant harmless from all such Taxes and all costs and expenses related thereto. When Consultant is acting as the merchant of record and City elects to include an additional fee in the End Users' cart that is identified as a

"sales tax" or similar designation, then, no more frequently than once per calendar year during the term of the Agreement, Consultant may, upon at least 5 business days' prior written notice, (i) require City to send to Consultant City's books and records related to its sales tax payments, and/or (ii) visit City's premises during City's normal business hours to review City's sales tax payments."

VII. Section 2.3, Reimbursable Expenses, is hereby renumbered to Section 2.4 and amended to read in its entirety as follows:

"2.3 2.4 Reimbursable Expenses. Compensation may include reimbursement for actual and necessary expenditures for reproduction costs, telephone expenses, and travel expenses approved by the Contract Officer in advance, or actual subcontractor expenses of an approved subcontractor pursuant to Section 4.5, and only if specified in the Schedule of Compensation. The Contract Sum shall include the attendance of Consultant at all project meetings reasonably deemed necessary by the City. Coordination of the performance of the work with City is a critical component of the services. Coordination with City as to the Products provided is a critical component of the Services. If Consultant is required to attend additional meetings to facilitate such coordination, Consultant shall not be entitled to any additional compensation for attending said meetings. The Contract Sum shall include the attendance of Consultant at all project meetings specified in the scope of work, and all meetings mutually deemed necessary by the City and Consultant."

VIII. Section 2.4, Invoices, is hereby reunumbered to Section 2.5 and amended to read in its entirety as follows:

"2.4 2.5 Invoices. Each month (except as otherwise provided in Section 2.2(a)), Consultant shall furnish to City an original invoice for all work Services performed and expenses incurred during the preceding month in a form approved by City's Director of Finance. By submitting an invoice for payment under this Agreement, Consultant is certifying compliance with all provisions of the Agreement. The invoice shall detail charges for all necessary and actual expenses by the following categories: labor (by sub-category), travel, materials, equipment, supplies, and subcontractor contracts. Subcontractor charges shall also be detailed by such categories. Consultant shall not invoice City for any duplicate services performed by more than one person all charges.

City shall independently review each invoice submitted by the Consultant to determine whether the work Services performed and expenses incurred Products provided are in compliance with the provisions of this Agreement. Except as to any charges for work performed or expenses incurred by Consultant which are disputed by City, or as provided in Section 7.3, City will use its best efforts to cause Consultant to be paid all amounts due (Fees will be paid as set forth in Section 2.2 and Exhibit "C") within forty-five (45) thirty (30) days of receipt of Consultant's correct and undisputed invoice; however, Consultant acknowledges and agrees that due to City warrant run procedures, the City cannot guarantee that

City's payment will occur within this time period. obligations will at times be delayed and fall outside of the thirty (30) day timeframe stated herein. In the event any charges or expenses are disputed by City, in good faith, the original invoice shall be returned by City to Consultant for correction and resubmission. Review and payment by City for any invoice provided by the Consultant shall not constitute a waiver of any rights or remedies provided herein or any applicable law."

- IX. Section 2.5, Waiver, is hereby renumbered to Section 2.6.
- X. A new Section 2.7, Fradulent Charges; Chargeback Requests; Refunds, is hereby added to the Agreement, to read in its entirety as follows:
 - "2.7 <u>Fraudelent Charges; Chargeback Requests; Refunds.</u> The Parties each hereby disclaim liability for fraudulent or unlawful charges attributable to third parties."
- XI. Section 3.1, Time of Essence, is hereby renamed and amended to read in its entirety as follows:
 - "3.1 Time of Essence Material Importance. Time is of the essence material importance in the performance of the Agreement."
- XII. Section 3.2, Schedule of Performance, is hereby amended to read in its entirety as follows:
 - "3.2 Schedule of Performance. Consultant shall commence the services Services pursuant to this Agreement upon receipt of a written notice to proceed and shall perform all services Services within the time period(s) established in the "Schedule of Performance" attached hereto as Exhibit "D" and incorporated herein by this reference. When requested by the Consultant, extensions to the time period(s) specified in the Schedule of Performance may be approved in writing by the Contract Officer but not exceeding one hundred eighty (180) days cumulatively."
- XIII. Section 3.4, Term, is hereby amended to read in its entirety as follows:
 - "3.4 <u>Term.</u> Unless earlier terminated in accordance with Article 7 of this Agreement, this Agreement shall continue in full force and effect until completion of the services but not exceeding for three one (31) years from the effective date hereof, except as otherwise provided in the Schedule of Performance (Exhibit "D"). At City's option, and upon execution of a written agreement between the Parties, the foregoing Term may be extended for up to three (3) two-year extension periods for each extension."
- XIV. Section 4.1, Representatives and Personnel of Consultant, is hereby amended to read in its entirety as follows:

"4.1 Representatives and Personnel of Consultant. The following principals of Consultant ("Principals") are hereby designated will designate a principal ("Principal") as being the principals and representatives representative of Consultant authorized to act in on its behalf with respect to the work Services specified herein and make all decisions in connection therewith:

(Name)	(Title)
(Name)	(Title)
(Name)	(Title)

It is expressly understood that the experience, knowledge, capability and reputation of the foregoing principals were a substantial inducement for City to enter into this Agreement. Therefore, the foregoing principals The Principal, which is subject to change over the course of this Agreement, shall be responsible during the term of this Agreement for directing all activities of Consultant and devoting sufficient time to personally supervise the services Services hereunder. All personnel of Consultant, and any authorized agents, shall at all times be under the exclusive direction and control of the Principals. For purposes of this Agreement, the foregoing Principals The Principal may not be replaced nor may their its responsibilities be substantially reduced by Consultant without the express written approval of City. Additionally, notice to City. Consultant shall utilize only competent personnel to perform services Services pursuant to this Agreement. Consultant shall make every reasonable effort to maintain the stability and continuity of Consultant's staff and subcontractors, if any, assigned to perform the services required under this Agreement. Consultant shall notify City of any changes in Consultant's staff and subcontractors, if any, assigned to perform the services required under this Agreement, prior to and during any such performance."

XV. Section 4.5, Prohibition Against Subcontracting or Assignment, is hereby amended to read in its entirety as follows:

"4.5 Prohibition Against Subcontracting—or; Assignment. The experience, knowledge, capability and reputation of Consultant, its principals and employees were was a substantial inducement for the City to enter into this Agreement. Therefore, Consultant shall not contract with any other entity to perform in whole or in part the services required hereunder without the express written approval of the City. In addition, neither this Agreement nor any interest herein may be transferred, assigned, conveyed, hypothecated or encumbered voluntarily or by operation of law, whether for the benefit of creditors or otherwise, without the prior written approval of City. Transfers restricted hereunder shall include the transfer to any person or group of persons acting in concert of more than twenty five percent (25%) of the present ownership and/or control of Consultant, taking all transfers into account on a cumulative basis. In the event of any such unapproved transfer, including any bankruptcy proceeding, this Agreement shall be void. No approved transfer shall release the Consultant or any surety of Consultant of any liability

hereunder without the express consent of City. Notwithstanding the foregoing, Consultant may assign its rights, claims, interests, or obligations under this Agreement without the consent of City to an affiliate company of Consultant or through any reorganization, merger, consolidation or asset sale of Consultant. Consultant shall make all reasonable efforts to notify the Contract Officer of any such reorganization, merger, consolidation or asset sale as soon as legally possible while accommodating the legal and business requirements of the transaction.

City may not resell, assign, or transfer any of its rights or obligations hereunder without Consultant's prior written approval, and any attempt to resell, assign, or transfer such rights or obligations without Consultant's prior written approval will be null and void."

- XVI. Section 5.1(d), Professional Liability, is hereby deleted in its entirety.
- XVII. Section 5.1(e), Subcontractors, is hereby amended to read in its entirety as follows:
 - "(e) <u>Subcontractors.</u> Consultant shall include all *any* subcontractors as insureds under its policies or shall furnish separate certificates and certified endorsements for each subcontractor. All coverages for subcontractors shall include all of the requirements stated herein."

XVIII. Section 5.1(f), Additional Insurance, is hereby deleted in its entirety and replaced with the following:

- "(f) Cyber Liability. Technology professional liability errors & omissions insurance appropriate to Consultant's profession and the Services hereunder with limits not less than \$5,000,000 per claim/loss, and \$5,000,000 in the aggregate. Coverage shall be sufficiently broad to respond to the duties and obligations undertaken by Consultant pursuant to this Agreement and shall include, but not be limited to, claims involving infringement of intellectual property, copyright, trademark, invasion of privacy violations, information or data breach, theft, loss, damage or misuse, release of private information, extortion and network security. If coverage is maintained on a claims-made basis, Consultant shall maintain such coverage for an additional period of three (3) years following termination of the contract. The insurance shall include the following coverage:
- i. Liability arising from the theft, dissemination and/or use of confidential or personally identifiable information; including credit monitoring and regulatory fines arising from such theft, dissemination or use of the confidential information.
- ii. Network security liability arising from the unauthorized use of, access to, or tampering with computer systems.

- iii. Liability arising from the failure of Consultant's proprietary technology products (software) required under the contract for Consultant to properly perform the services intended.
- iv. Electronic Media Liability arising from personal injury, plagiarism or misappropriation of ideas, domain name infringement or improper deep-linking or framing, and infringement or violation of intellectual property rights.
- v. Liability arising from the failure to render professional services as defined by industry standards appropriate to the technology being used."

XIX. Section 5.2, General Insurance Requirements, is hereby amended to read in its entirety as follows:

"5.2 General Insurance Requirements. All of the above policies of insurance shall be primary insurance and *the general liability and auto liability* shall name the City, its elected and appointed officers, employees and agents as additional insureds and any insurance maintained by City or its officers, employees or agents may apply in excess of, and not contribute with Consultant's insurance. The insurer is deemed hereof to waive all rights of subrogation *on the worker's compensation* and contribution it may have against the City, its officers, employees and agents and their respective insurers. Moreover, the insurance policy must specify that where the primary insured does not satisfy the self-insured retention, any additional insured may satisfy the self-insured retention.

All of said policies of insurance shall provide that said insurance may not be amended or cancelled by the insurer or any party hereto without providing thirty (30) days prior written notice by certified mail return receipt requested to the City. In the event any of said policies of insurance are cancelled, the Consultant shall, prior to the cancellation date, submit new evidence of insurance in conformance with Section 5.1 to the Contract Officer.

No work or services under this Agreement Services shall commence until the Consultant has provided the City with Certificates of Insurance, additional insured endorsement forms or appropriate insurance binders evidencing the above insurance coverages and said Certificates of Insurance or binders are approved by the City. City reserves the right to inspect complete, certified copies of and endorsements to all required insurance policies at any time. Any failure to comply with the reporting or other provisions of the policies including breaches or warranties shall not affect coverage provided to City.

All *The general liability and auto liability* certificates shall name the City as additional insured (providing the appropriate endorsement) and shall conform to the following "cancellation" notice:

CANCELLATION:

SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATED THEREOF, THE ISSUING COMPANY SHALL MAIL THIRTY (30)-DAY ADVANCE WRITTEN NOTICE TO CERTIFICATE HOLDER NAMED HEREIN.

[to be initialed]	
	Consultant Initials

City, its respective elected and appointed officers, directors, officials, employees, agents and volunteers are to be covered as additional insureds as respects: liability arising out of activities Consultant performs; products and completed operations of Consultant; premises owned, occupied or used by Consultant; or any automobiles owned, leased, hired or borrowed by Consultant. The coverage shall contain no special limitations on the scope of protection afforded to City, and their respective elected and appointed officers, officials, employees or volunteers. Consultant's insurance shall apply separately to each insured against whom claim is made or suit is brought, except with respect to the limits of the insurer's liability.

Any deductibles or self-insured retentions must be declared to and approved by City. At the option of City, either the insurer shall reduce or eliminate such deductibles or self-insured retentions as respects City or its respective elected or appointed officers, officials, employees and volunteers or the Consultant shall procure a bond guaranteeing payment of losses and related investigations, claim administration, defense expenses and claims. The Consultant agrees that the requirement to provide insurance shall not be construed as limiting in any way the extent to which the Consultant may be held responsible for the payment of damages to any persons or property resulting from the Consultant's activities or the activities of any person or persons for which the Consultant is otherwise responsible nor shall it limit the Consultant's indemnification liabilities as provided in Section 5.3.

In the event the Consultant subcontracts any portion of the work in compliance with Section 4.5 of this Agreement, the contract between the Consultant and such subcontractor shall require the subcontractor to maintain the same policies of insurance that the Consultant is required to maintain pursuant to Section 5.1, and such certificates and endorsements shall be provided to City."

XX. Section 5.3, Indemnification, is hereby amended to read in its entirety as follows:

"5.3 Indemnification. To the extent not prohibited by law, City will defend, indemnify, and hold Consultant harmless from and against any third party claim, demand, cause of action or proceeding (whether threatened, asserted, or filed) ("Claims") against Consultant to the extent that such Claim is based upon (i) injury or death to a person or damage to property resulting from such person's participation in an Event operated by City in connection with the Services and/or SaaS; (ii) City's provision of materials, products, or services, as part of City's obligations hereunder, that infringe the intellectual property rights of any third

party, provided that such materials, products, or services are used by Consultant in accordance with the Agreement; (iii) City's use, except as necessary to comply with applicable laws including but not limited to any disclosure under the Freedom of Information Act or the Public Records Act of Participant Information during the Term of this Agreement; (iv) unauthroized disclosure of Participant Information by a City employee or third party to whom access is given to Participant Information as provided hereunder, except for any disclosure that is required by applicable law; (v) City's use of the Services and/or SaaS in violation of this Agreement; or (vi) any claims for refunds, reversals or chargeback requests from End Users, except those created by a failure of Consultant to adhere to the provisions of this Agreement or applicable law.

To the full extent permitted not prohibited by law, Consultant agrees to indemnify, defend and hold harmless the City, its officers, employees and agents ("Indemnified Parties") against, and will hold and save them and each of them harmless from, any and all actions, either judicial, administrative, arbitration or regulatory claims, damages to persons or property, losses, costs, penalties, obligations, errors, omissions or liabilities whether actual or threatened (herein "claims or liabilities") that may be asserted or claimed by any person, firm or entity arising directly out of or in connection with the negligent performance of the work, operations or activities provided herein of Consultant, its officers, employees, agents, subcontractors, or invitees, or any individual or entity for which Consultant is legally liable ("indemnitors"), or arising from Consultant's or indemnitors' reckless or willful misconduct, or arising from Consultant's or indemnitors' negligent performance of or failure to perform any term, provision, covenant or condition of this Agreement, and in connection therewith:

- (a) Consultant will defend any action or actions filed in connection with any of said claims or liabilities and will pay all costs and expenses, including legal costs and attorneys' fees incurred in connection therewith;
- (b) Consultant will promptly pay any judgment rendered against the City, its officers, agents or employees for any such claims or liabilities arising out of or in connection with the negligent performance of or failure to perform such work, operations or activities of Consultant hereunder; the Services, and Consultant agrees to save and hold the City, its officers, agents, and employees harmless therefrom;
- (c) In the event the City, its officers, agents or employees is made a party to any action or proceeding filed or prosecuted against Consultant for such damages or other claims arising out of or in connection with the negligent performance of or failure to perform the work, operation or activities of Consultant hereunder, Services, Consultant agrees to pay to the City, its officers, agents or employees, any and all costs and expenses incurred by the City, its officers, agents or employees in such action or proceeding, including but not limited to, legal costs and attorneys' fees.

Consultant shall incorporate similar indemnity agreements with its subcontractors and if it fails to do so Consultant shall be fully responsible to indemnify City hereunder therefore therefor, and failure of City to monitor compliance with these provisions shall not be a waiver hereof. This indemnification includes claims or liabilities arising from any negligent or wrongful act, error or omission, or reckless or willful misconduct of Consultant in the performance of professional services hereunder. The provisions of this Section do not apply to claims or liabilities occurring as a result of City's sole negligence or willful acts or omissions, but, to the fullest extent permitted by law, shall apply to claims and liabilities resulting in part from City's negligence, except that design professionals' indemnity hereunder shall be limited to claims and liabilities arising out of the negligence, recklessness or willful misconduct of the design professional.. The indemnity obligation shall be binding on successors and assigns of Consultant and shall survive termination of this Agreement."

XXI. A new Section 5.4, Intellectual Property Indemnfication, is hereby added to the Agreement, to read in its entirety as follows:

"5.4 Intellectual Property Indemnification. Consultant agrees to defend, indemnify, and hold harmless the City and pay all damages (including reasonable attorneys' fees) relating to any third party claim, demand, cause of action, or proceedings (whether threatened, asserted, or filed) ("Claims") against City to the extent that such Claim is based upon Consultant's Products (excluding Third Party Products) directly infringing a United States patent, registered United States copyright, or registered United States trademark, provided that the Products are used in compliance with this Agreement. The indemnity obligation shall be binding on successors and assigns of Consultant and shall survive termination of this Agreement."

XXII. A new Section 5.5, Disclaimers and Limitation of Liability, is hereby added to the Agreement, to read in its entirety as follows:

"5.5 Disclaimers and Limitation of Liability. Consultant warrants that it will perform the services in a manner consistent with industry standards applicable to the performance thereof, that it will make commercially reasonable efforts to ensure that the Products function properly and continuously, and that it will respond in a timely and diligent manner to all complaints and requests submitted by the City in accordance with Consultant's "Service Package Advanced 7" package. Except as otherwise set forth herein and to the extent permitted by applicable law, City acknowledges and agrees that the products are provided on an "as is" and "as available" basis. The warranties, if any, set forth herein are limited to their express terms and are in lieu of, and Consultant expressly disclaims to the maximum extent permitted by law, all other warranties, express or implied, oral or written, including any (a) warranty that the products are error-free or "bug"-free, accurate, secure, or reliable; (b) warranty that the products will operate without interruption; (c) warranty that all errors will be corrected or that the products will comply with any law, rule, or regulation; (d) implied warranties of merchantability, fitness for a particular purpose, title, or non-infringement; (e) implied warranties arising from statute, course of dealing, course of performance, or usage of trade; and (f) warranty that the products will meet City's requirements. Neither party will be liable for indirect consequential, punitive, exemplary, or special damages (in contract, statute, tort, or otherwise), including damages for lost profits, cost of replacement services, lost data, or loss of use of information or services, which arise from circumstances beyond such party's reasonable control, whether or not such party has previously been advised of the possibility of such damages."

XXIII. Section 5.4, Sufficiency of Insurer, is hereby renumbered to Section 5.6.

XXIV. Section 6.1, Records, is hereby amended to read in its entirety as follows:

Records. Consultant shall keep, and require subcontractors to keep, such ledgers, books of accounts, invoices, vouchers, canceled checks, reports, studies or other documents relating to the disbursements charged to City and services performed hereunder (the "books and records"), as shall be necessary to perform the services required by this Agreement and enable the Contract Officer to evaluate the performance of such services. Any and all such documents shall be maintained in accordance with generally accepted accounting principles and shall be complete and detailed. The Contract Officer shall have full and free reasonable access to such books and records at all times upon request during normal business hours of City, including the right to inspect, copy, audit and make records and transcripts from such records. Such records shall be maintained for a period of three (3) years following completion of the services hereunder, and the City shall have access to such records in the event any audit is required. In the event of dissolution of Consultant's business, custody of the books and records may be given to City, and access shall be provided by Consultant's successor in interest. Notwithstanding the above, the. Consultant shall fully cooperate with the City in providing access to the books and records if a public records request is made and to the extent disclosure is required by law, including but not limited to the California Public Records Act.

When Consultant is acting as the merchant of record and City elects to include an additional fee in the End Users' cart that is identified as a "sales tax" or similar designation, then, no more frequently than once per calendar year during the term of the Agreement, Consultant may, upon at least 5 business days' prior written notice, (i) require City to send to Consultant copies of City's books and records related to its sales tax payments, and/or (ii) visit City's premises during City's normal business hours to review City's sales tax payments."

XXV. Section 6.2, Reports, is hereby deleted in its entirety and replaced with the following:

"6.2 <u>Reports.</u> As part of the Services and within the SaaS, Consultant shall provide City with access to a software called "Basecamp," in which Consultant shall provide the City (including as many stakeholders as the Contract Officer requests) with access to status and progress reports, notes, assignments, account ownership information, and other information relative to the Services and Products. "Basecamp" shall be accessible to the

City until thirty (30) days after the "Go Live" date. Prior to closing City access to "Basecamp," Active shall provide City with a zip file backup of the full "Basecamp" project to City."

XXVI. Section 7.1, California Law, is hereby renamed "Governing Law."

XXVII. Section 7.2, Disputes; Default, is hereby amended to read in its entirety as follows:

"7.2 <u>Disputes; Default.</u>

- In the event that Consultant is in default Default under the terms of this Agreement, the City shall not have any obligation or duty to continue compensating Consultant (whether by payment of compensation or entitlement to Fees) for any work services performed after the date of default Default. Instead, the City may give notice to Consultant of the default Default and the reasons for the default Default. The notice shall include the timeframe in which Consultant may cure the default Default. This timeframe is presumptively thirty (30) days, but may be extended, though not reduced, if circumstances warrant. During the period of time that Consultant is in default Default, the City shall hold all invoices and shall, when the default Default is cured, proceed with payment on the invoices. In the alternative, the City may, in its sole discretion, elect to pay some or all of the outstanding invoices during the period of default Default. If Consultant does not cure the default Default, the City may take necessary steps to terminate this Agreement under this Article Section 7.8. Any failure on the part of the City to give notice of the Consultant's default Default shall not be deemed to result in a waiver of the City's legal rights or any rights arising out of any provision of this Agreement. For the purposes of this section, Default is defined as any material violation of or failure to comply with the provisions of this Agreement. Consultant has represented that it will provide a 99.5% uptime for its Services 24x7 365 days per year. Should the actual uptime fall below 99.5%, such failure shall be deemed a Default and City shall be entitled to pro rata credits.
- (b) In the event that Consultant reasonably believes that City's use of the Services or SaaS constitutes a Default under this Agreement, Consultant may give the City written notice of the Default and the reasons for the Default. The notice shall include the timeframe in which Consultant may cure the Default. This timeframe is presumptively thirty (30) days, but may be extended, though not reduced, if circumstances warrant. If City does not cure the Default, Consultant may take necessary steps to terminate this Agreement under Section 7.8. Any failure on the part of Consultant to give notice of the City's Default shall not be deemed to result in a waiver of Consultant's legal rights or any rights arising out of any provision of this Agreement.
- (c) If Consultant reasonably believes that a transaction made using the SaaS or otherwise pursuant to this Agreement may be fraudulent or otherwise contrary to law, Consultant shall provide written notice to the City of the transaction and the reasons Consultant believes it is fraudulent or otherwise

contrary to law. Consultant shall maintain the ability to temporarily suspend services in the event of a suspected fraudulent event in accordance with Section 9.11, except that where the fraud is suspected to have been committed by an End User or other third party out of City's control, the cure provisions of Section 9.11 shall not apply, and the suspension shall be in effect for no longer than the minimum time period necessary to ensure the security of the SaaS and the City's account. City shall bear no responsibility for any damages or losses incurred by Consultant or any End User or other person or entity resulting from a fraudulent or unlawful transaction made pursuant to this Agreement unless the City committed the fraud or unlawful conduct underlying the transaction or the same is the result of the City's negligence or willful misconduct. Conversely, Consultant shall bear no responsibility for any damages or losses incurred by City or any End User or other person or entity resulting from a fraudulent or unlawful transaction made pursuant to this Agreement unless Consultant committed the fraud or unlawful conduct underlying the transaction or the same is the result of the Consultant's negligence or willful misconduct."

XXVIII. Section 7.3, Retention of Funds, is hereby deleted in its entirety and replaced with the following:

Retention of Funds. The Parties acknowledge that payments made by End **"7.3** Users to register for City Events using the SaaS pursuant to this Agreement will be initially received by Consultant as a merchant of record. Consultant shall convey all such Event registation fees, once received, to City's Finance Department on a weekly or bi-weekly basis. City hereby authorizes Consultant to charge Fees in accordance with Section III of Exhibit "C" an amount exceeding the standard Event registration fees such that the Fees charged will not in any way result in deduction of Event registration revenues to be reimitted to City. By way of example only, if the standard Event registration charge is \$100, any Fees charged by Consultant will result in City receiving \$100 even after Fees are charged and collected for any given transaction. Summary-level Fee reports shall be generated with each remittance cycle; the calculation and detailed support for all fees collected from End Users and fees disbursed by Consultant to City shall be available to the City as reports as set forth in Section II of Exhibit "A." In the event of a dispute, the Parties hereby agree to meet and confer in good faith to attempt to resolve the dispute without the need for formal legal action, and in the event such dispute is so resolved, the amounts previously withheld pursuant to the dispute shall be released on the next scheduled weekly or bi-weekly remittance date following resolution of the dispute. Except as otherwise stated in this section, Consultant's obligation to remit Event registration revenues or other monies due to the City shall be absolute and unconditional."

XXIX. Section 7.7, Termination Prior to Expiration of Term, is hereby amended to read in its entirety as follows:

"7.7 <u>Termination Prior to Expiration of Term.</u> This Section shall govern any termination of this Contract Agreement except as specifically provided in the

following Section for termination for cause. The City reserves the right to terminate this Contract at any time, with or without cause, upon thirty (30) sixty (60) days' written notice to Consultant, except that where and payment to Consultant of an early termination is due to fee in the fault of the Consultant, the period amount of notice may be such shorter time as may be determined by the Contract Officer. \$20,000. In addition, the Consultant reserves the right to terminate this Contract at any time, with or without cause, upon sixty (60) days' written notice to City, except that where termination is due to the fault of the City, the period of notice may be such shorter time as the Consultant may determine. Upon receipt of any notice of termination, Consultant shall immediately plan to cease all services hereunder except such as may be specifically approved by the Contract Officer. Except where the Consultant has initiated termination, the at the end of such sixty (60) day notice period. Consultant shall be entitled to compensation for all services rendered prior to the effective date of the notice of termination and for any services authorized by the Contract Officer thereafter in accordance with the Schedule of Compensation or such as may be approved by the Contract Officer, except as provided in Section 7.3. In the event the Consultant has initiated termination, the Consultant shall be entitled to compensation only for the reasonable value of the work product actually produced hereunder. In the event of termination without cause pursuant to this Section, the terminating party need not provide the non-terminating party with the opportunity to cure pursuant to Section 7.2."

XXX. Section 7.8, Termination for Default of Consultant, is hereby deleted in its entirety and replaced with the following:

Termination for Default. Either party may terminate this Agreement, immediately upon written notice: (a) in the event that the other party commits a non-remediable material breach of this Agreement, or fails to cure a Default pursuant to Section 7.2 or provide and comply with a mutually agreed-upon written plan to cure such Default; or (b) in the event of institution of bankruptcy, receivership, insolvency, reorganization, or other similar proceedings by or against either party under any section or chapter of the United States Bankruptcy Code, as amended, or under any similar laws or statutes of the United States or any state thereof, if such proceedings have not been dismissed or discharged within 30 days after they are instituted; or the insolvency or making of an assignment for the benefit of creditors or the admittance by either party of any involuntary debts as they mature or the institution of any reorganization arrangement or other readjustment of debt plan of either party not involving the United States Bankruptcy Code. Following termination of this Agreement, if requested by Consultant, City will certify that it has returned or destroyed all copies of the applicable Protected Materials, and acknowledges that its rights to use the same are relinquished, except to the extent any such materials were required to be produced pursuant to the California Public Records Act (in which case copies may be retained as necessary to comply with applicable record retention obligations), or are the subject of a dispute or litigation between the Parties pursuant to this Agreement. Termination for any reason will not excuse City's obligation to pay in full any and all amounts

due for services rendered prior to the effective date of termination, nor will termination by Consultant result in a refund of fees paid."

XXXI. Section 9.1, Notices, is hereby amended to read in its entirety as follows:

"9.1 Notices. Any notice, demand, request, document, consent, approval, or communication either party desires or is required to give to the other party or any other person shall be in writing and either served personally or sent by prepaid, first-class mail, in the case of the City, to the City Manager and to the attention of the Contract Officer (with her/his name and City title), City of Carson, 701 East Carson, Carson, California 90745 and in the case of the Consultant, to the person(s) attention of the Legal Department at the address designated on the execution page of this Agreement. Either party may change its address by notifying the other party of the change of address in writing. Notice shall be deemed communicated at the time personally delivered, the next day if sent via overnight mail, or in seventy-two (72) hours from the time of mailing if mailed as provided in this Section certified or registered."

XXXII. Section 9.2, Interpretation, is hereby amended to read in its entirety as follows:

"9.2 <u>Interpretation.</u> The terms of this Agreement shall be construed in accordance with the meaning of the language used and shall not be construed for or against either party by reason of the authorship of this Agreement or any other rule of construction which might otherwise apply. Any reference to a statutory provision includes a reference to any modification or re-enactment of it from time to time. The headings and pronouns contained herein are for convenience and ease of reference only and will not affect the construction or interpretation of this Agreement. The word "including" in this Agreement means "including, without limitation." All references to days means calendar days."

XXXIII. Section 9.3, Counterparts, is hereby amended to read in its entirety as follows:

"9.3 <u>Counterparts.</u> This Agreement may be executed in counterparts, each of which shall be deemed to be an *enforceable* original, and such counterparts shall constitute one and the same instrument. The parties agree that electronic or digital signatures, as well as pdf scanned copies of signatures, will be as effective and binding as original signatures."

XXXIV. Section 9.4, Integration; Amendment, is hereby amended to read in its entirety as follows:

"9.4 <u>Integration; Amendment.</u> This Agreement including the attachments hereto is the entire, complete and exclusive expression of the understanding of the parties. It is understood that there are no oral agreements between the parties hereto affecting this Agreement and this Agreement supersedes and cancels any and all previous negotiations, arrangements, agreements and understandings, if any, between the parties, and none shall be used to interpret this Agreement. No amendment to or modification of this Agreement shall be valid unless made in

writing and approved by the Consultant and by the City Council. The parties agree that this requirement for written modifications cannot be waived and that any attempted waiver shall be void. All pre-printed terms of any City or Consultant purchase order, schedule, business processing document, or on-line terms will have no effect. There have been no material representations or statements by any person or party to this Agreement as an inducement for a party hereto to accept this Agreement other than what is expressly set forth in writing herein."

XXXV. A new Section 9.8, Reservation of Rights, is hereby added to the Agreement, to read in its entirety as follows:

"9.8 Reservation of Rights. All rights not expressly granted in this Agreement are reserved by Consultant. City acknowledges that: (a) all Protected Materials are licensed and not sold; (b) City acquires only the right to use the Products in accordance with this Agreement, and Consultant and/or its licensors will retain sole and exclusive ownership of all rights, title, and interests in the Products, including the following: (i) all Intellectual Property embodied or associated with the Products, (ii) all deliverables and work product associated with the Products, and (iii) all copies and derivative works thereof; and (c) the Products, including the source and object codes, logic, and structure, contain and constitute valuable trade secrets of Consultant and/or its licensors."

XXXVI. A new Section 9.9, Restrictions, is hereby added to the Agreement, to read in its entirety as follows:

<u>Restrictions.</u> City will not: (a) sell, resell, distribute, host, lease, rent, license, or sublicense, in whole or in part, the Protected Materials; (b) decipher, decompile, disassemble, reverse assemble, modify, translate, reverse engineer, or otherwise attempt to derive source code, algorithms, tags, specifications, architecture, structure, or other elements of the Products in whole or in part, for competitive purposes or otherwise; (c) allow access to, provide, divulge, or make available the Protected Materials to any user other than those who are licensed to have such access; (d) write or develop any derivative works based upon the Products; (e) modify, adapt, translate, or otherwise make any changes to the Products or any part thereof; (f) use the Protected Materials to provide processing services to third parties, or otherwise use the same on a service bureau basis; (g) disclose or publish, without Consultant's prior written consent, and except as required by applicable law, including the California Public Records Act: (i) performance or capacity statistics, or the results of any benchmark test performed on the Products, or (ii) the terms (but not the existence) of this Agreement or other valuable trade secrets of Consultant; (h) without Consultant's prior written consent, and except as required by applicable law, perform or disclose or cause to be performed or disclosed any information related to any security penetration or similar tests; (i) disclose or otherwise use or copy the Protected Materials except as expressly permitted herein or as required applicable by law; (j) remove from any Products identification, patent, copyright, trademark, or other notices or circumvent or disable any security devices' functionality or features; (k) contest or do or aid

others in contesting or doing anything which impairs the validity of any proprietary or Intellectual Property rights, title, or interests of Consultant in and to any Products; (I) use the Products for other than authorized and legal purposes, consistent with all applicable laws, regulations, and the rights of others; (m) take any steps to avoid or defeat the purpose of security measures associated with the Products, such as sharing of login and password information, or attempt to circumvent any use restrictions; or (n) except as expressly permitted by this Agreement, use the Protected Materials for hosting purposes."

XXXVII. A new Section 9.10, Enforcement, is hereby added to the Agreement, to read in its entirety as follows:

"9.10 <u>Enforcement.</u> City will: (a) ensure that City's use (including any City contractors or anyone under control of City) of Products complies with the terms and conditions of this Agreement; (b) promptly notify Consultant of any actual or suspected violation thereof by a City official, agent, or employee; and (c) reasonably cooperate with Consultant with respect to any investigation and enforcement of any suspected use of the Products in violation of this Agreement."

XXXVIII. A new Section 9.11, Suspension, is hereby added to the Agreement, to read in its entirety as follows:

"9.11 Suspension. In accordance with this Section, Consultant may suspend its performance of services hereunder, including suspending remittance of payments and temporarily deactivating City's SaaS account, for a reasonable period of time, not to exceed thirty (30) days, in the event Consultant reasonably believes that City's use of the Services or SaaS is fraudulent or otherwise not in compliance with applicable law or this Agreement, or in the event Consultant reasonably and in good faith disputes the legal authority of City or any of its officers, agents, or employees to perform hereunder. Consultant shall provide 10 business days' written notice to City prior to effectuating any suspension, which notice shall describe the issue giving rise to the alleged basis for suspension in sufficient detail to enable the City to cure the issue, and the City shall have the opportunity to do so during the notice period. No suspension shall be effectuated if City cures the issue to the reasonable satisfaction of Consultant prior to expiration of the notice period. In the event a suspension is effectuated, and the nature of the suspension results in non-accrual of Fees during the suspension period, Consultant shall not be entitled to, and City shall not be liable for, any Fees that would otherwise have been generated or to which Consultant would have become entitled during such period had the suspension not been in effect."

XXXIX. A new Section 9.12, Export; Anti-Bribery, is hereby added to the Agreement, to read in its entirety as follows:

"9.12 <u>Export; Anti-Bribery.</u> The Products may include encryption software or other encryption technologies that may be controlled for import, transfer, export, or other purposes under Export Laws. City may not export, re-export, transfer, or re-

transfer or assist or facilitate in any manner the export, re-export, transfer, or retransfer of or provide access to any portion of the Products in violation of Export Laws, as determined by the laws of the State of California and Federal Law. City hereby represents and warrants that: (i) City is eligible to access the Products under Export Laws and all other applicable laws; and (ii) City will use or access the Products in accordance with Export Laws and all other applicable laws."

XL. A new Section 9.13, Relationship, is hereby added to the Agreement, to read in its entirety as follows:

"9.13 <u>Relationship.</u> This Agreement is not intended to create a partnership, franchise, joint venture, agency, or a fiduciary or employment relationship. Neither party may bind the other party or act in a manner which expresses or implies a relationship other than that of independent contractor."

XLI. A new Section 9.14, License to Intellectual Property/Promotion, is hereby added to the Agreement, to read in its entirety as follows:

"9.14 <u>License to Intellectual Property/Promotion.</u> Consultant hereby grants to City a limited, non-exclusive, non-transferable, non-sublicensable license during the term of this Agreement: (a) to use the SaaS for the purposes of offering, promoting, managing, tracking, and collecting fees in connection with City's Event(s) solely in accordance with this Agreement; and (b) to display, reproduce, distribute, and transmit in digital form Consultant's name and logo solely for the purposes set forth in this Section.

City hereby grants to Consultant a limited license to use information provided by City relating to City's organization and Events, which may include content regarding the Events, City's name, trademarks, service marks, and logo, seal, or insignia, in connection with the promotion of City's organization or Events or the Services that Consultant provides, provided that Consultant's use of or access to any such materials shall be subject to prior written approval of the Contract Officer on a case-by-case basis.

City will: (a) not use the SaaS to transmit, publish, or distribute any material or information: (i) for which City does not have all necessary rights and licenses, including any material or information that infringes, violates, or misappropriates the intellectual property rights of any third party; (ii) that contains a computer virus or other code, files, or programs designed to disrupt or interfere with the functioning of the SaaS; (iii) that is inaccurate or misleading; (iv) that is or that may reasonably be perceived as being harmful, threatening, offensive, obscene, or otherwise objectionable; (v) that contains a virus or malicious code; or (vi) that includes the private information of another without express permission, including but not limited to contact information, social security numbers, credit card numbers or other information which a reasonable person would consider private in nature; (b) not attempt to gain access to any systems or networks that connect to the Services and SaaS except for the express purpose of using the SaaS for their intended use;

(c) not engage in any activity that interferes with or disrupts the SaaS; (d) not use the SaaS in violation of applicable laws pertaining to unsolicited email, SMS, text messaging or other electronic communications."

XLII. A new Section 9.15, Information Collection; Account Access, is hereby added to the Agreement, to read in its entirety as follows:

"9.15 <u>Information Collection</u>; Account Access.

- (a) <u>Information Collection & Access.</u> Consultant collects Participant Information from End Users. Consultant represents and warrants that such collection does not violate the legal rights to privacy of the individuals whose information is collected, and agrees to defend, indemnify and hold harmless the City for claims or liabilities relating to violation of privacy rights in connection with collection of Participant Information, which agreement shall survive termination of this Agreement.
- (b) Account Access Notice; Covenant Not to Sue. The Contract Officer will immediately disable a user's access who is using the SaaS on the City's behalf or notify Consultant in writing if any such user is no longer authorized or is using such information without City's consent. Consultant may rely, without independent verification, on such notice, and City covenants not to sue Consultant for any claims arising from Consultant providing, denying, suspending, or modifying access to or use of the SaaS and Services of any individual as directed by the Contract Officer.
- (c) Account Ownership Disputes. In the event of any dispute between two or more parties as to end user account ownership, City agrees that Consultant will be the sole arbiter of such dispute in its reasonable discretion and that Consultant's decision (which may include termination or suspension of any end user account subject to dispute) will be final and binding on all parties. Restrictions. City agrees not to use the Software or Services to collect or elicit (a) any special categories of data (as defined in the California Consumer Privacy Act, as may be amended from time to time), including, but not limited to, data revealing racial or ethnic origin, political opinions, religious or other beliefs, trade-union membership, as well as personal data concerning health or sexual life or criminal convictions other than as expressly directed by Consultant, and in such event, only in predefined fields within the Software that are intended for that purpose; or (b) credit card information other than in pre-defined fields within the Software that are intended for that purpose.
- (d) <u>Use of Information Collected</u>. Both parties agree to use the collected information in compliance with: (a) all applicable laws, rules and regulations, including, without limitation, those governing privacy and personal information and the use of credit card data (e.g., using credit card information only for purposes authorized by the cardholder); (b) applicable credit card network rules and Payment Card Industry Data Security Standards; and (c) Consultant's privacy policy, as published on its website or otherwise provided by Consultant from time to time.

City will immediately disable a user's access who is using the SaaS on its behalf or notify Consultant in writing if any such user is no longer authorized or is using such information without City's consent. Consultant may rely, without independent verification, on such notice, and City, inclusive of City's parent, subsidiaries, and affiliated entities, as applicable, and each of their respective officers, directors, shareholders, owners, agents, employees, contractors, managers, representatives covenant not to sue and agree to defend, indemnify, and hold harmless Consultant from any claims arising from Consultant providing, denying, suspending, or modifying access to or use of the SaaS and Services of any individual as directed by City or by someone who Consultant reasonably, under the circumstances, believes is authorized to act on behalf of City. In the event of any dispute between two or more parties as to account ownership, City agrees that Consultant will be the sole arbiter of such dispute in its sole discretion and that Consultant's decision (which may include termination or suspension of any account subject to dispute) will be final and binding on all parties."

XLIII. A new Section 9.16, Existing Agreements; Account Access, is hereby added to the Agreement, to read in its entirety as follows:

"9.16 Existing Agreements. Nothing in this Agreement shall be construed to affect or interfere with any right or obligation of City or any other party under any City contract that is in effect as of the effective date hereof, including (without limitation) with respect to requiring use of Consultant's services in lieu of any City or third party event registration services provided for under such agreements."

XLIV. A new Section 9.17, Third Party Recipients; Account Access, is hereby added to the Agreement, to read in its entirety as follows:

"9.17 Third Party Recipients. City warrants that there will be no Third Party Recipients under this Agreement, provided Consultant complies with this section. Although the Parties acknowledge that City may use the SaaS for the benefit of a third party event, the Parties agree that under such circumstances, Consultant shall remit the event proceeds to City, and City shall subsequently be responsible for remitting such proceeds to the third party host, net of any monies (e.g., license/facility use fees) due to the City from such third party host."

XLV. A new Article 10, Definitions, is hereby added to the Agreement, to read in its entirety as follows:

"ARTICLE 10. DEFINITIONS.

- 10.1 "**Agreement**" means this Agreement, inclusive of all exhibits attached hereto and incorporated herein by reference, as defined in the preamble.
 - 10.2 "City" means the City of Carson, as defined in the preamble.

- 10.3 "Consultant" means Active Network, LLC, with a principal place of business at 717 N. Harwood St., Suite 2500, Dallas, TX, 75201, as defined in the preamble.
- 10.4 "**Desktop Software**" means each Consultant-developed and/or Consultant-owned software product in machine readable object code (not source code) that is installed on desktop(s) or server(s) controlled by City, the Documentation for such product, and any Updates and Upgrades thereto.
- 10.5 "**Documentation**" means the user instructions, release notes, manuals, or on-line help files in the form generally made available by Consultant, regarding the use of the applicable Software or Services, as updated by Consultant from time to time.
 - 10.6 "**Effective Date**" means the date set forth in the Preamble.
- 10.7 "**End Users**" means third party users who register for, sign up, or otherwise use the SaaS in connection with the Events, including reigstration therefor.
- 10.8 "Events" or "City Events" means City programs, facilities, events or memberships which are made available (for a cost or fee) to the public or City residents for reservation, registration or participation.
- 10.9 "**Export Laws**" means federal (U.S.) export control laws and regulations applicable to use, provision or access of the Products.
- 10.10 "**Hardware**" means computer hardware, equipment, and/or utilities supplied by Consultant pursuant to this Agreement.
- 10.11 "Intellectual Property" means any and all intellectual property and proprietary rights (in whole or in part) recognized by federal law, and whether or not perfected, filed, or recorded, including inventions, technology, patent rights (including patent applications, divisions, and disclosures), copyrights and all works of authorship (whether or not copyrightable), moral rights, trade secrets, trademarks and other indicators of source (and the goodwill associated therewith), service marks, trade dress, logos, methodologies, procedures, processes, know-how, tools, utilities, techniques, protocols, various concepts, ideas, methods, models, templates, software, source code, algorithms, tools, utilities, the generalized features of the structure, sequence and organization of software, user interfaces and screen designs, layouts, general purpose consulting and software tools, utilities and routines, and logic, coherence and methods of operation of systems, training methodology and materials, which Consultant has created, acquired, or otherwise has rights in, and may, in connection with the Products or the performance of Services hereunder, create, employ, provide, modify, create, acquire, or otherwise obtain rights in, and in each case includes any derivative works, alterations, and other modifications using, incorporating, based on, or derived from the foregoing.

- 10.12 "**Maintenance Services**" means the provision of Updates and Upgrades related to the Software all as more particularly set out in the applicable Exhibit to this Agreement.
- 10.13 "Participant Information" means information collected by Consultant from End Users pursuant to End Users' use of the SaaS. "Participant Information" does not include any information in the possession, custody or control of City, to the extent such information is obtained and held independently from City's use of the SaaS pursuant to this Agreement. Participant Information also includes any information input into the SaaS whether or not it was obtained previously or independently from the City's use of SaaS.
- 10.14 "**Products**" means, collectively, SaaS, Desktop Software, Services, Hardware, and all other services, products, or materials provided by Consultant to City under the terms of this Agreement.
- 10.15 "**Professional Services**" means the implementation, site planning, configuration, integration, and deployment of the Software or SaaS, training, project management, and other consulting services.
 - 10.16 "Protected Materials" means Products, except for Hardware.
- 10.17 "SaaS" means: (a) the software as a service which is hosted by Consultant or its hosting providers and which is accessed by City and its users via the internet; (b) Consultant's web sites; and (c) associated services, as more fully described in the Agreement. SaaS functionality is subject to change from time to time at Consultant's sole discretion.
- 10.18 "Services" means, collectively: (a) Professional Services; (b) Maintenance Services; (c) Support Services; and (d) any other services set forth in Exhibit "A." "Services" also includes the provision of Products.
 - 10.19 "**Software**" means the SaaS and the Desktop Software, collectively.
- 10.20 "**Support Services**" means the provision of technical assistance for Software or Hardware as further described in the Agreement and/or Exhibit "A."
- 10.21 "Taxes" means any and all applicable taxes, including sales, use, excise, withholding, assessments, stamp, transfer, value-added, duties, tariffs, export charges, import charges, and other taxes or assessments (however designated) imposed by any foreign, federal, provincial, state, or local governmental authority upon or applicable to Products arising out of this Agreement, other than those based on Consultant's net income."Third Party Products" means those hardware, firmware and/or software products, including updates and enhancements thereto, if any, owned by third parties, together with all user manuals and other documents accompanying the delivery of the Third Party Products.

- 10.22 "**Third Party Recipient**" means a third party that hosts an event for the benefit of which the City uses the SaaS, and which results in an obligation of Consultant to remit payment of proceeds from the event directly to the third party host.
- 10.23 "**Updates**" means bug fixes, patches, error corrections, minor releases, or modifications or revisions that enhance existing performance of the Software that are provided as part of Maintenance Services. Updates exclude Upgrades.
- 10.24 "**Upgrades**" means a new Software release that contains major functionality enhancements or improvements; and which is designated by an incremental increase in the release number to the left of the decimal point (by way of example only, release 5.0 designates an Upgrade from release 4.x). Upgrades exclude new products, modules or functionality for which Consultant generally charges a separate fee."

EXHIBIT "C" SCHEDULE OF COMPENSATION

I. Consultant shall perform the following tasks at the following rates:

TASK	UNIT PRICE	QUANTITY	SUB-BUDGET
1.	\$290.00	21	\$6,090.00*
2.	\$597.50	4	\$2,390.00*
3.	\$245.30	11	\$2,698.30*
4.	\$453.00	10	\$4,530.00*
5.	\$42,300.00	1	\$133,350.00**
6.	\$2,400.00	1	\$7,200.00***
7.	\$7,000.00	1	\$21,000.00***
8.	\$51,450.00	1	\$51,450.00*
9.	\$1,400.00	1	\$1,400.00*
10.	\$1,400.00	1	\$1,400.00*
11.	\$1,400.00	1	\$1,400.00*
SUBTOTAL:			\$232,908.30
SALES TAXES:			\$9,165.91
ADJUSTED TOTAL:			\$242,074.21
CONTINGENCIES (INCLUDES \$3,000 DATA EXPORT CHARGE AT END OF CONTRACT):			\$7,925.79
FINAL TOTAL:			\$250,000.00

^{.*}Amount reflects one-time cost during Year 1. No futher costs to be incurred by City for duration of Contract Term or any extended periods exercised by City under Section 3.4.

The Contract Sum is broken down by years as follows:

01007.0001/544964.15 C-1

^{**}The cost listed is the total cost to City during the Term. The annual cost to City is as follows: \$42,300.00 for Year 1, \$44,415.00 for Year 2, and \$46,635 for Year 3. The costs for Years 2 and 3 reflect cost of living increases. Additionally, for each year of any extended period of the Term, the cost is subject to a 0-5% cost of living increase, as determined by Consultant in its reasonable discretion.

^{***}Amount reflects total cost to City during the Term. The annual cost is the amount listed divided by 3.

Year 1 - \$127,006.34 plus \$2,993.66 for contingies = \$130,000.

Year 2 - \$56,371.21 plus \$3,628.79 for contingencies = \$60,000.00.

Year 3 - \$58,696.66 plus \$1,303.34 for contingies = \$60,000.00.

- II. Within the budgeted amounts in tasks 1-11 of Section I.I of this Exhibit, and with the approval of the Contract Officer, funds may be shifted from one task sub-budget to another so long as the Contract Sum is not exceeded per Section 2.1, unless Additional Services are approved per Section 1.8.
- III. During the Term of this Agreement, as consideration for the Services, Consultant shall be entitled to charge the following fees in connection with all transactions whereby members of the community pay for City Events using ACTIVE Net as provided pursuant to this Agreement (the "Fees"):

	FEE	FEE RATE
A.	Credit Card Processing Percentage Rate	3.00%
	(i.e., for every credit card transaction)	
	OR	
B.	Online Transaction Rate	3.00%
	(i.e., for every online transaction - convenience fee)	
C.	Additional Processing Fee	0.50%
	(i.e., staff facing electronic check payment)	

The Fee will either be 3% under Section III.A or a maximum of 3.5% under Sections III.B and III.C above.

In no event shall City be charged for any Fees.

50% of the total Professional Services fees will be billed at Service initiation and the remainder of the Professional Services fees will be billed upon completion of services. Both will be payable in accordance with Section 2.5.

Notwithstanding the foregoing, the total amount of Fees shall not exceed \$250,000 for the initial term of this Agreement. In the event City excercises any of its options to extend the term of this Agreement as set forth in Section 3.4, (i) the total amount of Fees for each year of any extension period shall not exceed 5% more than the Fees in effect for the immediately prior year, which amount is to be negotiated by the parties prior to any extension exercised pursuant to Section 3.4; and (ii) the parties shall negotiate the transaction Fee for each year of any extension period, it being understood that the Fees shall not exceed those set forth in Section III of this Exhibit.

01007.0001/544964.15 C-2

EXHIBIT "D" SCHEDULE OF PERFORMANCE

I. Consultant shall perform all Services timely in accordance with the following schedule:

Days to Perform	Deadline Date	
60 days	60 days from	
	Notice to	

<u>Tasks 5-7</u>:

Tasks 1-4:

N/A	N/A
(provided on an ongoing basis,	
throughout term of Agreement)	

Tasks 8-11:

Caalalarr	Caabalarr
See below	See below

Initiation	Project Launch	Remote
Week 1	Business Process Review	Remote
Week 3	Module 1-3 Review & Data Collection Preparation	Remote
Week 5	Module 4-7 Review & Data Collection Preparation	Remote
Week 7	Module ALL Data Sheet Review	Remote
Week 9	Module 1 & 2 User Testing	Remote
Week 11	Module 3 & 4 User Testing	Remote
Week 13	Module 5-7 User Testing	Remote
Week 15	Module ALL User Testing	Remote
Week 17	Training Plan Development	Remote
Week 19	LMS Core Training	N/A
Week 21	Supplemental Training	Remote
Week 23 & 25	Go Live Prep	Remote
Post Go Live	System Optimization Training	Remote

01007.0001/544964.15 D-1

II. Consultant shall deliver the following tangible work products to the City by the following dates.

See Scope of Services.

- III. The "Go Live" Date is contingent upon the Parties timely fulfilling their obligations under this Agreement as necessary to facilitate Consultant's efforts to "go live" by the agreed-upon deadline. If either Party fails to timely fulfill its obligations, the specified deadline shall be re-discussed and extended by a mutually agreeable time period reflecting the minimum time period reasonably necessary to account for the delay. Notwithstanding the foregoing or any other provision of this Agreement, Consultant's failure to timely satisfy any obligation in accordance with the applicable deadline or timeframe established pursuant to Section I.A within 15 calendar days of such deadline, unless an extension is approved in advance by the Contract Officer, shall constitute a non-remediable material breach by Consultant entitling City to (without limitation) terminate this Agreement in accordance with Section 7.8.
- IV. The Contract Officer may approve extensions for performance of the Services in accordance with Section 3.2.

01007.0001/544964.15 D-2